COLLECTIVE AGREEMENT

BETWEEN

THE SUDBURY CATHOLIC DISTRICT SCHOOL BOARD

AND

THE OECTA SECONDARY TEACHERS' LOCAL

(SUDBURY UNIT)

FOR THE

SCHOOL YEARS 2008-2009, 2009-2010, 2010-2011 & 2011-2012

Preamble	. 1
Article I - Recognition	. 1
1.01 - Exclusive Recognition	
1.02 - Application	
1.03 - <u>Definition of a Teacher</u>	
1.04 - Composition of the Local Bargaining Unit	
1.05 - Communication Between the Board and the Association	
1.06 - Commitment to Catholic Education	
	•
Article II - Duration	
2.01	. 3
Article III - Management Rights	. 3
3.01	
Article IV - Placement	1
4.01 - Evaluation of Qualifications	
4.02 - Experience	
4.02 - Experience	
•	
Article V - Salary Scales and Schedule of Allowances	
Preamble	
5.01 - Salary Scales	
5.02 - Method of Payment of Salaries	
5.03 - Transportation Allowance	
5.04 - Conference Allowance	
5.05 - Responsibility Allowance	. 7
Article VI - Benefits	. 7
6.01 - Extended Health Care Plan	
6.02 - Dental Plan	
6.03 - Group Term Life Insurance Plan	
6.04 - Long Term Disability Plan	
6.05 - Workplace Safety Insurance Plan	
6.06 - Sick Leave Credit System	
6.07 - Teachers on Leaves of Absence Without Salary	
6.08 - Early Retirement Incentive Plan	
v	

6.09 - Liability Insurance
6.10 - Retiree Benefits
6.11 - Annual Deposit
6.12 - Sick Leave Credit Gratuity
Article VII - Grievance Procedure
7.01 - Definitions
7.02 - Solution of Problems Discussed at the Administrative Level 19
7.03 - Procedure for Individual Grievances
STEP I
STEP II
STEP III
7.04 - Procedure for General and Group Grievance
STEP I
STEP II
7.05 - Arbitration
7.06 - Expedited Arbitration
7.07 - Time Limits
7.08 - No Reprisals
Article VIII - Seniority Lists
8.01 - Seniority Defined
8.02 - Seniority List Established
8.03 - Criteria for the Establishment of the Seniority List 25
8.04 - Day School Continuing Education Seniority List 26
8.05 -Criteria for the Establishment of the Day School Continuing
Education Seniority List
8.06 - Right to determine Number of Religious Teachers 27
Article IX - Staffing Formula
9.01 - Determining Requirements
9.02 - Staffing for Alternative School
9.03 - Sharing of Staffing Information
9.04 - Joint Board Staffing Committee (JBSC)
Article X - Organizational Units and Positions of Responsibility
10.01-Organization Unit Structure and Allocation
Article XI - Staffing Process

11.01 - Redundancy Determined	. 32
11.02 - Surplus to the School	. 33
11.03 - Posting	. 36
11.04 - Voluntary Specific Site and Term Transfer Request	. 37
11.05 - Transfer of Staff Members without their Expressed Request	. 38
11.06 - Procedure for Staff Reduction	. 38
11.07 - Right of Recall	. 39
11.08 - Appointment of Teachers to Positions of Responsibility and	
Allowances	. 40
11.09 - Teachers Employed by the Board	. 43
11.10 - Posting of Teaching Positions and Positions of Responsibility	. 43
11.11 - Job Sharing	. 44
Article XII - Leaves of Absence	
12.01 - Sabbatical Leave	
12.02 - Leave for Personal Reasons	. 46
12.03 - Leave of Absence due to Lengthy Illness	
12.04 -Leave of Absence (One year, one semester, part-time leave of	
absence)	. 47
12.05- Leave of Absence for Unit President and/or Unit Release Office	<u>r(s)</u>
of OECTA	. 49
12.06 - Deferred Salary Leave	. 50
12.07 - Pregnancy/Parental Leave (Including Adoption)	. 52
12.08 - Special Time Off	. 54
12.09 - Bereavement Leave	
12.10 - Special Leaves of Absence	. 56
12.11 - Return from Leave	. 56
12.12 - Special Leave Incentive Plan	. 56
12.13 - Family Medical Leave	
Article XIII - Teaching Conditions	
13.01 - Teaching Load	
13.02 - Personnel Files	
13.03 - Performance Appraisal	
13.04 - Offence Declarations	
13.05 - Credit Rescue/Recovery	
13.06 - Joint Professional Development Committee	
13.07 - Faculty of Education Student Placement	
13.08 - Credit Integrity	. 62

Article XIV - Supervision	62
14.01 - Uninterrupted Lunch	62
14.02 - On call Periods, Supervision Periods and Preparation Time	63
Article XV - Other Matters	64
15.01 - <u>Association</u> Fee	64
15.02 - Local Levy	64
15.03 - Printing and Distribution of this Agreement	64
<u>15.04</u> - Just Cause	64
15.05 - Board Statement to Teachers	65
15.06 - Continuing Education	65
15.07 - The Adult Continuing Education Program	66
15.08 - Reopener Clause	67
15.09 - Letters of Understanding - Letters of Intent	67
15.10 - Freedom from Assault in the Workplace	67
15.11 - Denominational Rights	67
<u>15.12</u> - Certified Teachers	68
15.13 - Temporary Administrative Assignments	68
15.14 - Replacing Teachers	68
15.15 - Union Representatives	69
15.16 - Trial Administrative Assignment	69
15.17 - Access to Information - Association	69
<u>15.18</u> - <u>Home Instruction</u>	69
SCHEDULE 'A'	7 1
LETTER OF INTENT	
School Year	
Posting Process for New Schools	74
Teachers Assigned to Non-Credit Programs	75
Adult Continuing Education Program	76
Access to Relevant Board Administrative Procedures and Guidelines	
E-Learning Credit Courses	78
Best Practices	79
LETTER OF UNDERSTANDING	
No Hiring of Non Members	80
Staffing (PDT)	81

PDT Benefits and Enhancements	82
Member Education on Leaves	83
APPENDIX 1	84

Preamble

WHEREAS, it is the common goal of the Board and the O.E.C.T.A. Secondary Teachers Local Bargaining Unit to provide the best possible Catholic educational services to the students of this community;

AND, WHEREAS, to achieve that common goal it is essential that the Board and the O.E.C.T.A. Secondary Teachers' Local Bargaining Unit maintain the harmonious relationship which exists between them;

AND, WHEREAS, the Board and the OECTA Secondary Teachers' Local Bargaining Unit are committed to improve student achievement, reduce gaps in student outcomes and increase confidence in publicly funded education;

THEREFORE it is the desire of the Board and the O.E.C.T.A. Secondary Teachers' Local Bargaining Unit to set forth in this agreement the salaries, allowances, benefits and any term or condition of employment as mutually agreed upon.

Article I - Recognition

1.01 - Exclusive Recognition

Pursuant to the Education Act as amended and the Regulations of the Ministry of Education and Training, the Sudbury Catholic District School Board (hereinafter called the Board) recognizes the O.E.C.T.A. Secondary Teachers' Local Bargaining Unit as the regular, sole and exclusive negotiating agent for all teachers comprising its membership.

1.02 - Application

The terms of this agreement shall apply to all members of the O.E.C.T.A. Secondary Teachers' Local Bargaining Unit.

1.03 - Definition of a Teacher

"Teacher" shall mean a "teacher as defined in the Education Act Part X.1, S.277.1: Part X.1 teacher means a teacher employed by the Board to teach but does not include a supervisory officer, a principal, a vice-principal or an instructor in a teacher-training

institution.

Persons for whom the Board has secured a Letter of Permission under Regulation 183 shall be deemed as teachers. The Board shall report to the Association every vacancy and appointment made under Regulation 183.

1.04 - Composition of the Local Bargaining Unit

The O.E.C.T.A. Secondary Teachers' Local Bargaining Unit shall be composed of every Part X.1 teacher who is assigned to one or more secondary schools operated by the Board, or to perform teaching duties in respect of such schools all or most of the time excluding Occasional Teachers.

Only OECTA Secondary members will be assigned to deliver the non-credit portion of Guidance.

1.05 - Communication Between the Board and the Association

All communication between the Board and the Association will be done through the Director of Education or designate and the President of the O.E.C.T.A. Secondary Teachers' Local Bargaining Unit.

1.06 - Commitment to Catholic Education

The Board and the Association value the publicly funded Catholic School system, and share responsibility for providing a Catholic education.

In recognition of the importance of a Catholic education system, both parties acknowledge the benefits of teacher professional development in the area of Catholic faith development. Any teacher who became a member of the OECTA Secondary Teachers Local Bargaining Unit effective September 1st, 2008, or thereafter, shall register and successfully complete the Religious Education Part 1 Additional Qualification Course operated by OECTA and OCSTA. Teachers must successfully complete the course within twenty-four (24) months of their date of hire. Within sixty (60) days of the successful completion of the course and submission of proof of payment, the teacher shall be reimbursed for one-half (1/2) of the cost of the course. An extension may be granted for extenuating circumstances at the sole discretion of the Superintendent of Employee Relations; however, reimbursements are only applicable to courses completed within the first twenty-four months of employment.

In adherence to the philosophy of Catholic Education, employees of the Sudbury Catholic District School Board are encouraged to direct their school taxes to the English Catholic school system.

Article II - Duration

2.01

- (1) This agreement shall be in effect on the first day of September, <u>2008</u> and continue in force until the 31st day of August, <u>2012</u> and shall continue automatically thereafter for annual periods of one year unless either party notifies the other in writing within 90 days prior to the expiration date that they desire to negotiate with a view to renewal, with or without modifications of this agreement, in accordance with the Ontario Labour Relations Act.
- (2) Notwithstanding the period of notice cited in (1), either party may notify the other in writing, within the period commencing April 1 prior to the expiration date that it desires to negotiate with a view to renewal, with or without modifications of this Agreement, in accordance with the Ontario Labour Relations Act.
- (3) If either party gives notice of its desire to negotiate amendments in accordance with Section (2), the parties shall meet within fifteen (15) working days from the giving of notice to commence negotiations for the renewal of the Agreement in accordance with the Ontario Labour Relations Act.
- (4) No changes can be made to this Agreement without the mutual written consent of the parties; nor can any changes be made to this Agreement without submitting the changes for ratification by the parties, as determined by their respective bargaining procedures.
- (5) No teacher shall be requested or required to perform the duties of any other teacher or employee of the Board who is engaged in a strike.

Article III - Management Rights

3.01

(1) It is the sole and exclusive right and obligation of the Board to exercise its

management functions and trustee responsibilities and to manage the affairs of the Board and to exercise these rights and obligations in a manner consistent with the collective agreement and subject to the provisions of the Ontario Labour Relations Act, the Constitution Act, 1867, and in particular Section 93 thereof, and the Acts and regulations of the Ministry of Education and the Province of Ontario.

- (2) A teacher will be considered to be on probation until the teacher:
 - a) has completed two years of employment with the Board, if the teacher has less than one year of experience, as a teacher in a secondary school in Ontario before commencement of employment with the Board; or
 - b) has completed one year of employment with the Board, if the teacher has one or more years of experience, as a teacher in a secondary school in Ontario before commencement of employment with the Board.
- (3) It is expected that teachers will provide two months notice of resignation for the purpose of retirement as verified by the Teachers Pension Plan.

Article IV - Placement

4.01 - Evaluation of Qualifications

- (1) The placement of each teacher shall be in accordance with the statements of evaluation issued by the Qualifications Evaluation Council of Ontario (QECO) based on Programme 5.
- (2) Each statement shall be accompanied by the original QECO covering letter which describes the details of the rating. The original letter will be returned to the teacher.

4.02 - Experience

(1) a) Experience shall mean the time recognized for remuneration purposes consisting of all properly documented teaching experience gained subsequent to graduation from a Teachers' College or Faculty of Education while teaching on an occasional basis or on a continuous basis while employed as a Teacher with a school board constituted under the

Education Act or an educational institution acceptable to the Board subject to paragraph (2), and shall be recognized in full to the nearest month as of September 1st of each year.

- b) Experience shall accumulate during all leaves for which salary is maintained and for the statutory duration of Family Medical leave, pregnancy leave and/or parental leave as defined in the Employment Standards Act of Ontario; it will be suspended during all other leaves.
- (2) With respect to teachers of Business or Technical Studies with or without an Ontario Teacher's Certificate, their related trade, industrial or business experience will be recognized as provided below.

Effective September 1, 2000 teachers who hold a Certificate of Qualification in a trade under the Trades Qualification and Apprentice Act and who possess College of Teachers qualifications to teach technological studies courses will be granted one year of teaching experience for salary purposes for each year of trade experience related to the subject to be taught and earned after trade certification but prior to employment with the Board. The courses eligible for the allowance are: Construction Technology, Manufacturing Technology, Technological Design, Transportation Technology, Communication Technology. The Board will determine the amount of trade experience to be credited.

4.03 - Implementation

- (1) A teacher who qualifies for a salary adjustment prior to September 1st in any year, will be entitled to a salary adjustment effective September 1st if he/she submits the acknowledgement of receipt of his/her application to QECO on or prior to December 31st and the Board receives the teacher's Certificate of Qualifications and revised QECO evaluation on or prior to March 1st of the following calendar year.
- (2) A teacher who qualifies for a salary adjustment after September 1st but on or prior to December 31st, will be entitled to a salary adjustment effective January 1st if he/she submits the acknowledgement of receipt of his/her application to QECO prior to March 1st and the Board receives the teacher's Certificate of Qualifications and revised QECO evaluation on or prior to May 1st of the same calendar year.

- (3) No request for a salary adjustment will be accepted after the last day in February.
- (4) The Board will confirm electronically the receipt of a teacher's Certificate of Qualifications and revised QECO evaluation certificate.
- (5) For the purpose of this clause, documents must be received by the Board or postmarked no later than the date specified in paragraphs (1) and (2) above.

Article V - Salary Scales and Schedule of Allowances

Preamble

The annual salary of each teacher shall be determined according to the following salary scales and schedule of responsibility allowances.

In the case of a teacher who is teaching less than full-time the annual salary shall be prorated according to the percentage of teaching time.

5.01 - Salary Scales

See Schedule "A".

5.02 - Method of Payment of Salaries

Teachers annual salary will be disbursed in equal bi-weekly installments on Thursday from September to August.

For teachers teaching semester 1 only, a wrap-up will be paid out as a lump sum at the end of the semester worked. For teachers teaching semester 2 only, an adjustment will be made on the bi-weekly payments in order to enable an equalized distribution of salary over the period of work and July and August.

A teacher who is not teaching a full school year is entitled to be paid his/her salary in the proportion that the total number of school days for which he/she performs his duties in the school year bears to the total number of school days in the school year.

5.03 - Transportation Allowance

Where the Board requests teachers to attend in-service meetings, or to travel during

the school day, the Board shall reimburse them at the greater of the rate of 35.0 cents per kilometre or as specified in Board <u>Administrative Procedures and Guidelines</u>.

5.04 - Conference Allowance

- (1) Where the Board requests teachers to attend special workshops, meetings, conferences, conventions or courses (excluding additional qualifications courses, such as Religious Education Part I, II and III operated by OECTA and OCSTA) outside the District of Sudbury as Board representatives, the Board shall reimburse the teacher's expenses upon receiving the appropriate receipts as required for:
 - a) registration fee
 - b) transportation costs as approved by the Director of Education or designate
 - c) accommodation costs up to \$150.00 per day upon submission of receipts
 - d) per diem of \$50.00 (receipts not required)
- (2) Where attendance at any one of the above functions is at the teacher's request, the Board may reimburse part or all of the expenses at the discretion of the Director of Education or designate.
- (3) The Board shall attempt to replace any teacher referred to in paragraph (1) with an occasional teacher.

5.05 - Responsibility Allowance

Responsibility allowance shall be paid to heads of organizational units in accordance with Article XI (Staffing Process).

Article VI - Benefits

In the case of part-time teachers, the Board's contribution for benefits under clauses 6.01 and 6.02 will be prorated according to the percentage of teaching time as specified in the contract of employment.

The parties to this agreement acknowledge that the cost increase of premiums will be calculated as part of the overall costs to the Board for negotiation purposes.

The Board shall provide the OECTA <u>Secondary Local Bargaining Unit</u> with the <u>premium</u> costs, <u>enrolment levels and utilization of the benefits</u> for the <u>bargaining</u> unit, on an annual basis within 15 days of being received by the Board.

The Board shall provide the OECTA Sudbury Secondary Local Bargaining Unit with the Benefits' Plans Master Policies in effect and the updated Master Policies, following amendment or renewal, within 15 days of being received by the Board. The parties acknowledge and recognize that the Board is not the insurer; however the Board will maintain same or equivalent coverage in the Master Policies fo the life of this agreement. However, it is understood that any enhancements to benefits as a result of the PDT Benefits Enhancements Letter of Understanding shall be incorporated effective September 1, 2010.

The Local Bargaining Unit shall be consulted during the process in the event of a change in carriers.

The Board shall ensure that all members of the OECTA Sudbury Secondary Local Bargaining Unit shall have access to information regarding benefits coverage. The Board shall consult with the President of the Bargaining Unit regarding the content and the method of delivery of this information and shall endeavor to make the information available in a timely manner but no later than March 1, 2009.

6.01 - Extended Health Care Plan

The Board shall pay 100% of the premiums for the Comprehensive Extended Health Care Plan in place as at September 1, 2002 which includes a \$250 vision care plan every two (2) years, a \$2.00 deductible pay-direct drug plan, which will only cover prescription drugs and includes mandatory generic substitution with physician override, a semi-private plan and \$50,000.00 per year Private Duty Nursing.

6.02 - Dental Plan

- (1) The Board shall pay 100% of the premium for a dental plan which limits examinations (check ups) to every twelve (12) months for adults and every six (6) months for dependent children.
- (2) This plan will reflect the current ODA schedule in effect.
- (3) This plan will include a 50% co-insurance orthodontal and major restorative

plan to a maximum of \$2,000. lifetime for each element for all family members.

6.03 - Group Term Life Insurance Plan

- (1) The Board shall provide group term life insurance equal to two (2) times the annual <u>salary</u> taken to the next higher \$ 1,000, pro-rated to the percentage of teaching time. In addition, the Board shall provide spousal term life insurance coverage of \$18,000. The Board shall pay 100% of the <u>group life and the spousal insurance premiums</u>. Participation in this insurance plan shall be mandatory for all Teachers covered by this agreement.
- (2) Teachers will have the option to purchase additional coverage (Optional Life Insurance) equivalent to one (1) time the annual salary taken to the next higher \$1,000 at the Board's premium rate. Coverage is subject to insurability by the carrier. This additional coverage will be 100% at the teacher's expense.
- (3) For Life Insurance coverage purposes, annual salary includes allowances as per clause 11.08 (1) Program Leader, clause 11.08 (2) Coordinator/Consultant, clause 15.16 Trial Administrative Assignment and the additional President's allowance outlined in clause 12.05 (3).

6.04 - Long Term Disability Plan

- (1) The Board shall administer a Long Term Disability Insurance Plan selected by the Local Bargaining Unit of OECTA only through the collection and transfer of premiums to the insurance company and the completion of the employer's statement on the disability claim form submitted by Teachers. Participation in this plan shall be mandatory except for any teachers who elected to opt out of the Plan prior to Sept 1, 2003.
- (2) To facilitate the early identification and timely processing of LTD claims, the Board agrees to notify the Association President when an individual teacher's absence reaches fifteen (15) consecutive days.
- (3) Teachers shall pay 100% of the premiums for the plan and shall have the sole right to determine the carrier. The Association shall provide required documentation on the plan to the Board. In the event of a change in carrier the Association shall notify the Board at the earliest opportunity and no later than 120 days prior to such change.

(4) Article 12.03(2) and 6.07 apply to teachers on LTD.

6.05 - Workplace Safety Insurance Plan

A teacher on staff who is injured in the course of his/her duty and receiving indemnity from the Workplace Safety Insurance Board (W.S.I.B.) shall be provided the opportunity to elect:

- (1) to receive his/her indemnity directly from the W.S.I.B. without deduction from his/her accumulated sick leave credits_and in such case, subject to Section 25 of the Workplace Safety Insurance Act, may participate in group benefits provided in this agreement by remitting directly to the Board the full premium thereof, or
- (2) to receive from the Board his/her regular gross salary, consisting of W.S.I.B. award (advances) plus a top-up amount; the Board shall then deduct from the employee's cumulative sick leave bank, for each day of absence, the proportion of time_equivalent to the top-up (salary less W.S.I.B. award) paid by the Board.
 - When the Teacher's cumulative sick leave credits are entirely used up, the conditions of option (1) above shall apply.
- (3) The Board shall provide the Local Bargaining Unit with a copy of the employer's report of the injury or disease (Form 7) when submitting such form to the Workplace Safety and Insurance Board with the teacher's authorization.
- (4) The Board agrees to provide to the Local Bargaining Unit and the injured teacher any return to work plan or any other prescribed information and/or correspondence between the Board and the W.S.I.B. regarding a teacher's W.S.I.B. claim with the teacher's authorization.

6.06 - Sick Leave Credit System

(1) Subject to subsection (2), a sick leave credit system is hereby established, based on twenty (20) sick leave days per year, for every teacher covered by this agreement. The Board shall, on September 1, of each school year, credit each teacher with twenty (20) days sick leave. In the case of a part-time teacher the amount of sick leave credits allowed per month will be prorated to the percentage of the week worked

- (2) Effective September 1, 1998, sick leave credits shall be granted to a newly hired teacher on the basis of two (2) days per month for the first year of the teacher's employment.
- (3) The Director of Education or his/her designate shall administer the sick leave credit plan.
- (4) The Human Resources Department shall keep a register or registers in which shall be entered the credits, the accumulated credits and the deduction therefrom.
- (5) All teachers may accumulate unused sick leave credits.
- (6) Calculations resulting in fractions of less than one-half day throughout this plan shall be adjusted upwards to the nearest half-day.
- (7) One hundred percent of the unused portion of his/her sick leave shall be transferred to the credit of each Teacher each year as his/her accumulated sick leave credit to a maximum of 240 days.
- (8) Deduction from the Accumulated Sick Leave Credit Reserve

The sick leave credit reserve or any part thereof may be used for sick leave for any one year in addition to the sick leave credit of twenty days. In calculating the amount by which the sick leave absence reduces the Teacher's sick leave, such absences shall be deducted from the annual sick leave credit first, then when that has been used up, the accumulated sick leave credit reserve shall be drawn upon.

- (9) a) Every Teacher will notify the Board's representative or agency as designated and his/her principal of his/her absence and the probable date of his/her return.
 - b) All claims for a period of more than three (3) consecutive days, whether that claim be for annual sick leave credit or for sick leave credit reserve, must be properly supported by a doctor's certificate to be in the hands of the Board within ten (10) consecutive days following the commencement of sick leave or upon release from the hospital, if the Teacher is hospitalized.

- whenever requested by the Board to do so, a doctor's certificate at the Board's expense to support his/her claim and the Board at its discretion shall have the right to have the Teacher_examined by a doctor selected by the Board. Factors that the Board shall consider in selecting a doctor, shall include the Teacher's preference for the gender of the doctor, if expressed.
- (10) All payments to Teachers under the plan shall be computed on the basis of the rate of the regular salary such Teacher is, or would be receiving at the time the absence occurs.
- (11) A teacher with a pregnancy related illness may use Article 6.06(8). A teacher with pregnancy related illness following delivery may use Article 6.06(8) for up to 30 working days.
- (12) Where a teacher has used the twenty (20) day annual leave and the teacher's entire accumulated sick leave, the Director of Education may use his/her discretion in regards to payment of salary if the teacher requires additional sick leave.
- (13) When a teacher employed by a School Board that has established a sick leave credit plan is hired by the Board, the Board shall place to the credit of the teacher the sick leave credits standing to the credit of the teacher in the plan of the first mentioned Board subject to the limit of 240 days in the Board's plan, as required by Section 180 of the *Education Act*.

6.07 - Teachers on Leaves of Absence Without Salary

Teachers on leaves of absence without salary (excluding pregnancy/parental and Family Medical leave) shall be permitted to remain with the various group plans as outlined in the contract but with the provision that the teacher will be responsible for 100% of the cost of the premiums. The full amount of the premiums for such coverage will be paid monthly by automatic bank withdrawals authorized by (voided) cheque.

6.08 - Early Retirement Incentive Plan

Preamble

The provisions of this article shall apply only during a time of projected teacher redundancies and, in all cases, the total number of applications allowed shall not exceed the number of redundant teachers declared under Article 11.01 herein.

- (1) A teacher who is eligible to retire on a reduced superannuation pension may elect to retire prior to reaching the age of 65 years.
- (2) Eligibility is restricted to teachers
 - a) with a minimum of 10 years experience with the Board,
 - b) who are at the maximum in their salary category, and
 - c) who submit an irrevocable application to the Board by April 10th for retirement effective between June 30th and August 31st of the same year and December 10th for retirement effective January of the following year.
- (3) A maximum of 3 teachers shall be eligible for the plan.
- (4) Approval for requests shall be governed by seniority.
- (5) Access to the plan shall be limited to retirements on or before the <u>expiration of this collective agreement.</u>
- (6) The financial incentive shall be 5% of the annual salary per year of early retirement to a maximum of 25% of annual salary on the date of retirement.
- (7) The early retirement incentive shall be paid in full within one year after retirement or as arranged to the mutual satisfaction of the teacher and the Board upon submission by the teacher to the Human Resources Department of written confirmation of the granting of a pension by the Teacher's Pension Plan Board.

6.09 - Liability Insurance

The Board shall continue to effect and keep in force an adequate policy or policies of insurance, insuring each teacher in its employ when acting in the course of such teacher's employment, including any Board approved activity organized by the school, in or out of school hours, against liability in respect of any claim for damages or personal injury.

6.10 - Retiree Benefits

- (1) For current and future retired employees who have had 15 or more years of continuous service with the Board, the Board agrees to contribute 100% of the total premium cost for the following plans until such employees attain 65 years of age.
 - a) Comprehensive Extended Health Care which includes the semi-private plan and a prescription drug plan with annual deductible of \$25.00 single and \$50.00 family with eye glass subsidy of \$150.00 every two (2) years.
 - b) The provisions of this article apply only to a person who:
 - i) has applied within ten years of normal retirement for a service pension from the Teacher Pension Plan Board; or
 - ii) has applied for a Teacher Pension Plan Board Disability Pension.
- (2) Teachers retired from teaching with the Board may continue to have access to the group coverage plans under articles 6.01, 6.02 and 6.03. The full amount of the provisions for such coverage will be paid by monthly automatic bank withdrawals authorized by (voided) cheque. It is understood that retired employees form a separate group with respect to plan administration, experience and premium schedules.
- (3) The surviving spouse of a deceased teacher, including a retired teacher, shall be entitled to continue coverage under 6.01 and 6.02 as the case may be, in the Board's group insurance plans for a period not exceeding the earliest of the date on which the teacher would have reached the age of 65 years, the date on which the surviving spouse reaches the age of 65 years or the date on which the surviving spouse remarries. It is understood that a surviving spouse who maintains coverage will form part of the retired teacher group with respect to plan administration, experience and premium schedules. It is further understood that the surviving spouse will pay the full cost of premiums by monthly automatic bank withdrawals authorized by (voided) cheque.

6.11 - Annual Deposit

(1) Any teacher covered under Article 1.03 of this Collective Agreement, may receive benefits either under this Article or under Article 6.12. (Sick Leave Credit Gratuity) Participation in either plan has no effect on the Sick Leave

Credit System 6.06.

- (2) a) A teacher may choose to receive benefits under this Article in lieu of Article 6.12 at anytime up to the first ten (10) years of service however once the employee opts for the Annual Deposit Plan, the choice is final and may not be reversed.
 - b) Each year employees shall be sent an election form by <u>December 31</u> in order to apply by <u>April 30</u> to opt for the Annual Deposit Plan. To assist teachers in making their selection, a personalized data sheet comparing the plans shall be included by the Board.
- (3) a) The Board shall contribute a lump sum calculated as the number of years of Board Experience in a position covered by this collective agreement multiplied by the annual contribution rate to the maximum contribution (refer to contribution schedule).
 - b) A teacher who receives a lump sum that is less than the maximum contribution will continue to receive annual contributions as per the contribution schedule until the maximum contribution is reached.
 - c) The contribution shall be made on August 31 following the date of the teacher's request and on August 31 every year thereafter (maximum 10 years) following the completion of a minimum of five (5) months of teaching in any one school year.
- (4) The annual/maximum contribution amount is determined by the number of years of Board Experience the employee has as of August 31st at the end of the school year in which the employee opts for the Annual Deposit Plan.

Contribution Schedule

Board Experience	Board Contribution	
Years — (as of Aug. 31st)	Annual	Maximum
0.0 to 3.9	\$700.00	\$7,000.00
4.0 to 5.9	900.00	9,000.00

6.0 to 7.9	1,000.00	10,000.00
8.0 to 9.9	1,100.00	11,000.00
10.0+	1,200.00	12,000.00

- (5) A teacher who resigns from the employment of the Board and who is later rehired will receive contributions under paragraph (3) for the difference between the amount previously contributed by the Board and the maximum contribution, if any. Such teacher will not be entitled to additional contributions under this Article and will not be entitled to benefits under Article 6.12.
- (6) The Board shall deposit any contribution required under this Article into an account designated by the teacher and held at a chartered bank, credit union or caise populaire.
- (7) The Board shall assist employees who participate in the Annual Deposit Plan by establishing a payment schedule that optimizes their RRSP contribution room. Any payments withheld for such purpose shall be paid with interest at the rate of interest earned by the Board
- (8) Prior to any teacher choosing to accept an annual deposit under this clause, the teacher shall consult with the Local Bargaining Unit and indicate their intention in writing.
- (9) The Board shall provide to the Local Bargaining Unit the list of employees who have opted for the annual deposit plan each semester.
- (10) In the event of the death of a teacher before receiving the full amount of the lifetime contribution, the annual contribution amount plus any amount held by the Board on behalf of the employee for the current year will be paid to the teacher's estate.

6.12 - Sick Leave Credit Gratuity

(a) A teacher qualified under paragraph (b) to receive a sick leave gratuity will receive a gratuity as follows:

10 years service, 10% of cumulative sick leave credit (with a maximum accumulation of sick leave credits of 200 days) x 1/200th of annual salary at date of retirement from teaching:

```
11 years service, 12% x 1/200th of annual salary
12 years service, 14% x 1/200th of annual salary
13 years service, 16% x 1/200th of annual salary
14 years service, 18% x 1/200th of annual salary
15 years service, 20% x 1/200th of annual salary
16 years service, 22% x 1/200th of annual salary
17 years service, 24% x 1/200th of annual salary
18 years service, 26% x 1/200th of annual salary
19 years service, 28% x 1/200th of annual salary
20 years service, 30% x 1/200th of annual salary
21 years service, 32% x 1/200th of annual salary
22 years service, 34% x 1/200th of annual salary
23 years service, 36% x 1/200th of annual salary
24 years service, 38% x 1/200th of annual salary
25 years service, 40% x 1/200th of annual salary
26 years service, 42% x 1/200th of annual salary
27 years service, 44% x 1/200th of annual salary
28 years service, 46% x 1/200th of annual salary
29 years service, 48% x 1/200th of annual salary
30 years service, 50% x 1/200th of annual salary
```

Partial years of service shall be rounded to the nearest month. For each month of service over the number of completed years of service 0.2% shall be added to the % provided for in this paragraph.

- (b) Saving any rights which the teacher may have acquired prior to this date and by virtue of Section 180, subsection 1 of the Education Act, R.S.O. 1990, c.E.2., and amendments thereto, after ten (10) years service with the Board:
 - i) the teacher who is retired from the Board, and qualifies for, and is in receipt of a pension from the Ontario Teachers' Pension Plan Board;
 - ii) the teacher who retires from the Board and who elects to transfer the commuted value of their pension to another retirement savings arrangement in accordance with the provisions of the Teachers' Pension

Act will be deemed to have retired on pension for the purpose of qualifying for a retirement gratuity if their date of retirement is within one (1) year of the date they would have qualified to receive an unreduced pension from the Teachers' Pension Plan;

- iii) Where the Director of Education permits at his/her discretion on compassionate grounds, the teacher who retires from the Board and/or who elects to transfer the commuted value of their pension to another retirement savings arrangement in accordance with the provisions of the Teachers' Pension Act will be deemed to have retired on pension for the purpose of qualifying for a retirement gratuity, and the Director's decision will not be grievable.
- (c) In the event of the death of a teacher either before or after retirement but before recovering the full benefits of the accumulated sick leave as provided under subparagraph (a), such remaining benefits shall be paid to the teacher's estate.
- (d) All benefits provided under subparagraph (a) shall be paid in full within one year after retirement or as arranged to the mutual satisfaction of the teacher and the Board upon submission by the teacher to the Manager of Human Resources or designate, of written confirmation of the granting of a pension or a commuted value transfer by the Ontario Teachers' Pension Plan Board.

Article VII - Grievance Procedure

7.01 - Definitions

- (1) A grievance shall be defined as any difference arising out of the interpretation, application, administration or alleged violation of this agreement, and is identified as one of the following:
 - a) An individual grievance is a grievance lodged by or on behalf of one covered by this collective agreement, or
 - b) A group grievance is a grievance lodged on behalf of two or more members covered by this collective agreement by the Local Bargaining Unit upon a common issue, or lodged by the Board against two or more members covered by this collective agreement upon a common issue, or,
 - c) A general grievance is a grievance lodged by either party, other than

under a) or b) above.

- (2) A party is:
 - a) the Bargaining Unit;
 - b) The Board.
- (3) A member is a teacher in the employ of the Board and who is covered by this collective agreement.
- (4) Days shall mean business days unless otherwise stated.

7.02 - Solution of Problems Discussed at the Administrative Level

- (1) It is felt that most problems can be solved satisfactorily at the administrative level by contacting the Superintendent of Education or designate.
- (2) The parties may proceed to a process of consultation within five (5) days of the initial contact made in 7.02(1) whereby a meeting may be arranged which shall include a member of administration, the member involved, the President or designate of the Local Bargaining Unit and any other persons who may be of assistance in arriving at a satisfactory resolution to the problem.

7.03 - Procedure for Individual Grievances

STEP I

- (1) In the case of an individual grievance, the member or Local Bargaining Unit initiating the grievance shall make a written statement containing:
 - a) a description of how the alleged dispute is grievable as defined in clause 7.01 (1) (a);
 - b) the circumstances giving rise to the grievance, and the matter complained of;
 - c) the relief sought; and
 - d) the signature of the grieving member and of the duly authorized official of the Local Bargaining Unit, when applicable;

and deliver same to the Superintendent of Education or designate within fifteen (15) days of the meeting, under clause 7.02, should the Superintendent of Education have been contacted within fifteen (15) days of the occurrence.

(2) The Superintendent of Education or designate shall reply in writing within 10 days following receipt of the signed grievance under paragraph (1) above. If prior to such reply, a meeting is desired by either the grieving member or the Board, such meeting shall be held between the grieving member who may be accompanied by a representative of OECTA and the Superintendent of Education or designate who likewise may be accompanied by another Board designate.

STEP II

- (1) If a satisfactory settlement is not reached at Step I, the grieving member or OECTA, with the grieving member's written consent, may request within ten (10) days of receipt of the reply of the Superintendent of Education or designate, that the grievance be referred to the Director of Education for a meeting. This meeting will be held within ten (10) days of receipt of such request, at which time the grieving member may be accompanied by a representative of OECTA who may attend to bring evidence and discuss the matter.
- (2) The Director of Education shall render his/her decision in writing within ten (10) days following the meeting.

STEP III

If the reply of the Director of Education is not satisfactory, OECTA may, within ten (10) days of the receipt of the reply, request in writing that the matter be referred to arbitration for adjudication. In situations where the employment relationship is in jeopardy, OECTA may request, in writing, that the matter be referred to the Committee of the Whole Board.

If a satisfactory settlement is not reached at STEP I, OECTA may decide to refer the matter immediately to arbitration for adjudication.

7.04 - Procedure for General and Group Grievance

The following steps shall be taken by a party in the case of a general or group

grievance. A general or group grievance may be filed by either the Local Bargaining Unit or the Board and either party may decide at which step the procedure will begin.

A general grievance or group grievance shall be initiated within 30 calendar days from the date of the occurrence giving rise to such grievance.

STEP I

The party initiating the grievance shall make a written statement containing:

- (1) a description of how the alleged dispute is grievable as defined in clause 7.01 (1) (b) or (c);
- (2) the circumstances giving rise to the grievance, the matter complained of and the name of the teacher(s) involved;
- (3) the relief sought; and
- (4) the signature of the duly authorized official of the party making the grievance; and deliver same
 - a) to the Superintendent of Education or designate
 - b) to the President of the Local Bargaining Unit

as the case may be, who shall, within 10 days of receipt of same, reply in writing.

STEP II

If the reply of the Superintendent of Education or designate or the President of the Local Bargaining Unit is not acceptable, the party making the grievance may within ten (10) days of receiving such reply, request in writing that the matter be referred to arbitration for adjudication.

The timeline for requesting arbitration may, by mutual agreement, be extended if both parties agree to refer the matter to the Director of Education in the interim who shall provide a written reply within ten (10) days following a meeting at which both parties shall have been heard. In such circumstances, if the reply made is not acceptable, the party making the grievance may, within ten (10) days of receiving such reply, request in writing that the matter be referred to arbitration for adjudication.

7.05 - Arbitration

- (1) The party desiring Arbitration shall, within the time lines above, notify the other party in writing both of its desire to submit the grievance to arbitration and of its choice to submit the matter either to a single arbitrator or to a three-member Arbitration Board. The notice will also contain the name of the first party's nominee as single Arbitrator or its appointee to a three-member Arbitration Board. The recipient of the notice shall, within ten (10) days of receipt of the notice, inform the other party:
 - a) where a single Arbitrator has been requested, either that it accepts the other party's nominee or of the name of its own nominee.
 - b) where a three-member Board has been requested, the name of its appointee to such Board.

If single arbitration has been selected but the parties fail to agree within ten (10) days on the nominee, the appointment shall be made by the Minister of Labour on the request of either party. If the appointees to an Arbitration Board fail to agree upon a Chair within ten (10) days, the appointment shall be made by the Minister of Labour at the request of either party. The single Arbitrator or the three-member Board, shall hear representations by the parties and/or representatives and determine the grievance and shall issue a decision and the decision shall be final and binding upon the parties and upon any member or members affected by it. The decision of majority is the decision of the Arbitration Board, but, if there is no majority, the decision of the Chair governs.

- (2) The Arbitrator or three-member Board shall not by decision, add to, delete from, modify or otherwise amend the provisions of the Agreement.
- (3) It is anticipated that the single Arbitrator or three-member Board will make every effort to render its written decision thirty days from the date of the completion of the hearing of the grievance.

(4) Cost of Arbitration:

The fees for a single Arbitrator, or a chairperson of a three-member Board, shall be shared equally by the parties. Each party shall bear the cost of its own appointee on a three-member Arbitration Board.

7.06 - Expedited Arbitration

- (1) Notwithstanding the procedure above, either party may request access to Expedited Arbitration under Section 49 of the *Ontario Labour Relations Act*.
- (2) The names of the following Arbitrators shall serve as Chairs for a panel of arbitrators for the duration of the agreement:

Richard Brown Nimal Dissanayake Robert Howe Brian Keller Susan Tacon

- (3) The parties are to canvass the above Arbitrators in rotation, beginning with the name following that of the last Arbitrator agreed upon, to determine the first Arbitrator who has an available date within 30 days to act as Chair of a Board of Arbitration. It is the responsibility of the parties to secure the attendance of their own nominee and counsel on the specified day.
- (4) Additional names may be added to the panel of arbitrators by mutual consent of the both parties to this agreement.

7.07 - Time Limits

Time limits may be extended if mutually agreed upon in writing. If the grievor fails to comply with the time limits, the grievance shall be deemed abandoned. If the respondent fails to comply with the time limits, the grievor shall be at liberty to enter the grievance in the next succeeding stage. Forwarding of the required documents by registered mail, <u>Board electronic mail</u> or delivery by hand to the party's representative within set time limits shall be considered as complying with the time limits. Receipt of a document shall be on the <u>next business</u> day <u>after</u> it is delivered, if hand delivered <u>or sent via Board e-mail</u> to a party or on the third <u>business</u> day following its postmark, if forwarded by registered mail.

7.08 - No Reprisals

There shall be no reprisals of any kind taken against any person because of participation in a grievance or arbitration procedure under this Agreement.

Article VIII - Seniority Lists

8.01 - Seniority Defined

- (1) As of September 1, 2001, seniority shall, for the purpose of this agreement in establishing priority among teachers who are employed by the Board and teaching at the secondary level with the Board, mean the length of time since the most recent effective start date as stated in the letter of offer of employment.
- (2) Seniority accumulated in the Local Bargaining Unit with the Sudbury Catholic District School Board prior to September 1, 2001 shall be recognized on the basis of the formula and seniority list applicable on August 31, 2001.
- (3) For the purpose of the tie breaker, "continuous employment" shall mean the continuous employment with the Board since the most recent effective start date in the Local Bargaining Unit where "continuous employment" shall include:
 - a) exchange teaching
 - b) loan to DND
 - c) Federation leaves
 - d) any and all leaves taken with the approval of the Board
 - e) leaves for lengthy illness

8.02 - Seniority List Established

- (1) Each teacher employed by the Board and teaching at the secondary level with the Board shall be placed on the seniority list. If a teacher is not employed full-time, the appropriate fraction will be indicated on the seniority list.
- (2) The Seniority List shall be established by the Board in consultation with the Local Bargaining Unit.
- (3) The Seniority List shall be continuously updated with a copy thereof provided to the Local Bargaining Unit October 30, January 31, and April 30.
- (4) Part-time teachers shall not be pro-rated.

8.03 - Criteria for the Establishment of the Seniority List

(1) Seniority shall be established within Grade 9 to Grade 12 on the basis of the most recent effective start date with the Board in the Local Bargaining Unit as per Article 8.01. The seniority list shall be based on the length of time since the

most recent effective start date with the Board in the Local Bargaining Unit.

- 2) Should there be a tie in seniority, the following shall be used as a tie breaker in descending order:
 - a) where the teachers have the same seniority, the length of continuous employment with the Board from the the most recent effective start date with the Board as defined in article 8.01 shall be the first tie breaker to establish ranking;
 - b) where the teachers have the same seniority under (2) (a) the order on the list shall be determined on the basis of total teaching experience with the Board or its predecessor boards; then,,
 - c) where teachers have the same seniority under (2) (a) and (b) the order on the list shall be decided upon the basis of total teaching experience in Ontario; then,
 - d) where teachers have the same seniority under (2) (a) (b) and (c) the order on the list shall be decided upon on the basis of total teaching experience in Canada; then,
 - e) where teachers have the same seniority under (2) (a) (b) (c) and (d) the order on the list shall be decided on the basis of QECO rating; then
 - f) where teachers have the same seniority under (2) (a), (b), (c), (d) and (e) the order on the list shall be decided by lot drawn in the presence of the President of the Local Bargaining Unit or designate
- (3) Secondary school teachers hired by the Board from the Sudbury Board of Education due to the financing of Catholic secondary schools will be credited all of the seniority they have accumulated while in the employ of the Sudbury Board of Education.

8.04 - Day School Continuing Education Seniority List

- (1) Each Continuing Education teacher teaching Day School Continuing Education credit courses shall be placed on a seniority list.
- (2) The seniority list shall be established by the Board in consultation with the Local Bargaining Unit.

- (3) The seniority list shall be updated each year as of September 30th and a revised copy thereof provided to the Local Bargaining Unit as of October 30th.
- (4) Effective September 1, 1997, criteria for determining the ranking of teachers on the seniority list will be the most recent continuous teaching experience with this Board as defined in Article $8.0\underline{5}(1)$.
- (5) Part-time teachers shall not be pro-rated

8.05 - Criteria for the Establishment of the Day School Continuing Education Seniority List

- (1) Seniority shall be established within the Day School Continuing Education component of the Board as follows:
 - a) the seniority list shall consist of the names of teachers in decreasing order of years of continuous employment;
 - b) where the teachers have the same length of continuous employment with the Board from the effective date of employment, the order on the list shall be determined on the basis of total teaching experience with the Board or its predecessor boards; then,
 - c) where teachers have the same seniority under (1) a) and b) the order on the list shall be decided upon on the basis of total teaching experience in Ontario; then,
 - d) where teachers have the same seniority under (1) a), b) and c) the order on the list shall be decided upon on the basis of total teaching experience in Canada; then,
 - e) where teachers have the same seniority under (1) a), b), c) and d) the order on the list shall be decided on the basis of QECO rating; then,
 - f) where teachers have the same seniority under (1) a), b), c), d) and e) the order on the list shall be decided by lot drawn in the presence of the President of the Local Bargaining Unit or designate.

8.06 - Right to determine Number of Religious Teachers

In order to maintain classes at the Secondary Level in cooperation with religious

orders, it is accepted that the Board in consultation with the religious order of each institution will have the right to determine yearly the number of religious teachers to be employed in Grades 9 to 12.

Article IX - Staffing Formula

9.01 - Determining Requirements

- (1) Staffing projections at <u>each school</u> will be determined based on March 31st projected enrollments. The full year staffing complement <u>at each school</u> will be calculated at 98% of the March 31st projected enrollment. Staffing adjustments <u>at each school</u> will be made on the second Friday of the new school year using 98% of the actual enrolment numbers on that date, <u>in consultation</u> with the President of the Local Bargaining Unit and/or the Joint Board Staffing Committee. All information pertinent to determining staffing allocations shall be shared by both parties.
- (2) The <u>minimum</u> number of F.T.E. classroom teachers required <u>at each school</u>, except for the <u>Self-Contained/Special Education/Life Skills/Comprehensive classes</u>, shall be calculated according to the following formula:
 - (i) <u>A x C</u> B x D

Where:

A = Number of regular program F.T.E. students at 98%

B = Average class size as regulated by the Ministry of Education

C = Per pupil credit load to a maximum as regulated by the Ministry of Education

D = Teaching load as regulated by the Ministry of Education (6 of 8)

- (a) The target number of students per section in the Locally Developed/Essential programs will be sixteen (16) students.
- (ii) Resource Allocation:

Special Education Resource teachers will be assigned on the basis of 1.00 FTE Resource teacher for each 750 regular program students (excluding students in Self-Contained/Special Education/Life Skills/Comprehensive classes).

(iii) Self-Contained/Special Education Allocation:

- (a) Special Education class teachers will be assigned on the basis of 1.00 F.T.E. teacher for each full-time special education class (Life Skills, Developmental, Comprehensive).
- (b) The target number of students per section in the comprehensive sections will be sixteen (16) students.
- (c) The target number of students per section in the life skills, developmental class will be ten (10) students.

(iv) Student Success Allocation

- a) The target number of students per section in the student success programs/learning strategies/credit recovery will be sixteen (16) students.
- b) The number of Student Success Teachers assigned board wide to secondary schools shall be consistent with the Ministry of Education funding and allocation provided for this purpose.
- c) The Student Success Teacher(s) at each secondary school shall have a timetable that designates teaching periods as per Article XIII.
- d) The Board shall send to the Local Bargaining Unit president a copy of the Annual Student Success Action Plan.
- e) Prior to the new school year the student success teacher allocation at each school will be reviewed with the Local Bargaining Unit.

(v) Guidance Allocation:

There shall be five (5) Guidance teachers assigned in the system. Assignments to schools will be determined by the Superintendent of Education in consultation with the school principals. The assignment shall be reviewed by the Local Bargaining Unit.

(vi) Additional Programming, (e.g. Native Studies, Alternative Schools Programs for Expelled and Suspended Students)

The number of teachers assigned Board wide for these initiatives will be based on the Ministry funding provided for this purpose.

- (vii) The sum of staff allocations in Article 9.01 <u>for each school</u> shall be rounded up to the nearest fraction of an FTE teacher.
- (3) There shall be <u>two (2)</u> teacher librarian assigned in the system.
- (4) Additional teachers per school to be allocated by the Principal in consultation with the Superintendent of Education.

9.02 - Staffing for Alternative Programs

- (1) Alternative programs shall include, but are not limited to, Dual-Credit Programs, School-College-Work Initiatives, E-Learning, Specialist High Skills Majors Programs, and Programs for Expelled and Suspended Secondary Students.
- (2) Secondary teachers shall be assigned, as per Article XI, at each secondary school, to alternative programs in consultation with the Joint Board Staffing Committee. Such assignments shall appear on teacher timetables and classes shall appear on student timetables.
- (3) All secondary school credits are granted by the Principal based on the evaluation of a member of the Secondary Teachers Local Bargaining Unit notwithstanding where they are delivered.
- (4) A secondary school's Average Daily Enrolment in "Dual Credit" courses shall be included in the calculation of the number of secondary teaching positions required in the Board pursuant to this Collective Agreement and/or any class-size regulation.
- (5) Should the Board decide to assign the designation of "position of responsibility" to teachers in the Alternative School/Program(s), the remuneration shall be as per the Program Leader's compensation in Article 11.08 (5).
- (6) The delivery of programs to students who have been suspended for a duration of six (6) to ten (10) and eleven (11) to twenty (20) instructional days and to students who have been expelled and who elect to participate in the Alternative School Program shall be the responsibility of the teachers assigned to the position of Alternative School/Program(s) for expelled and suspended students.

9.03 - Sharing of Staffing Information

The President of the Local <u>Bargaining</u> Unit and the Superintendent of <u>Employee</u>

<u>Relations or designate</u> will meet <u>no later than October 1 and March 1 of each school year to review class size and teacher assignments</u>.

9.04 - Joint Board Staffing Committee (JBSC)

The Parties agree that staffing needs are to be addressed in a Joint Board Staffing Committee (the "JBSC"). Within ninety (90) days of ratification of this collective agreement, the JBSC shall be established in order to provide a higher level of openness, disclosure and meaningful consultation.

- a) The JBSC shall be composed of the three (3) representatives selected by the Board and three (3) representatives selected by Secondary Local Bargaining Unit and mutually agreed appropriate resource staff.
- b) Meetings shall be chaired alternately by the Board and the Local Bargaining Unit. Once the JBSC is established it shall be convened no later than September 30th, in each school year, for an initial meeting. Thereafter, the committee shall meet five (5) times annually, unless otherwise agreed to by the parties. An agenda for each meeting shall be prepared prior to any meeting.
- c) The Terms of Reference for the JBSC shall include, but not be limited to:
 - i) Monitoring the existing staffing model and consulting on the staffing process;
 - ii) Reviewing and discussing secondary school staffing data and making recommendations to the Director of Education/designate on the deployment of teachers to individual schools from staff allocated to the system;
 - iii) Addressing other staffing and workload issues;
 - iv) Reviewing, sharing and consulting on the staffing and implementation of programs that include but are not limited to: Adult Continuing Education Program, Dual Credits, Specialist High Skills Majors, E-learning, Native Studies Programs, Credit Recovery, Alternative Education/Program for Expelled and Suspended Students, Laptop Program, and Student Success; and
 - v) Consulting on such other staffing issues as the JBSC agrees to.

- vi) Sharing all relevant reports and data related to staffing and workload in a timely manner.
- d) In the event that the JBSC is unable to agree on a matter within its mandate, or to reach a decision on a timely basis, the provisions of this Collective Agreement otherwise apply.

The parties agree that the JBSC shall work in an atmosphere of collegiality, openness and transparency and to further Board-level labour relations.

Article X - Organizational Units and Positions of Responsibility

10.01 - Organization Unit Structure and Allocation

- (1) Organizational Units in each secondary school, with the exception of St. Albert's Adult Learning Centre, will be led by a teacher designated as a Program Leader.
- (2) The complement of secondary school program leader positions shall be as follows: effective September 2008 seventeen (17), effective September 2009 eighteen (18). Effective September 2010, the number of Program Leader positions to specific schools shall be based on the ratio of one (1) Program Leader for every one-hundred-eleven (111) students to a maximum of nineteen (19). The allocation of these positions to specific schools shall be based on the representation by population formula. These determinations shall be made by the Board by April 15.
- (3) a) The curriculum areas assigned to each Organizational Unit shall be chosen from the following list:

Religion Modern Languages

English Music Mathematics Art

Social Science Guidance

Science Technological Studies
Business Computer Studies
Special Education Family Studies

Drama

Physical and Health Education (Academic/Co-Curricular, Extra Curricular)

b) Each school must have within one of its Organizational Units "Religion"

as part of the title.

- c) Each curriculum area listed above shall be assigned to an Organizational Unit in every school.
- (4) Prior to <u>April 30th</u>, the Principal, in consultation with the staff and the <u>President</u> of the Local Bargaining Unit will prepare a plan of Organizational Units outlining the curriculum areas in each Organizational Unit.

Article XI - Staffing Process

11.01 - Redundancy Determined

- (1) Redundant teachers shall be those in excess of projected staffing requirements as determined by Article IX and recognizing the need for reasonable flexibility in determining the number of redundant staff.
- (2) Redundant positions will be determined on a system-wide basis from Grade 9 to 12 by April 15.

(3) Procedure

- a) Teachers will be declared redundant strictly in order of reverse seniority.
- b) Teachers to be declared redundant will be notified in writing by April 15. The letter will state that the reason for the termination of the teacher's employment is a decrease in enrolment. The letter will also inform redundant teachers of the provisions of this Collective Agreement that could lead to their re-employment.
- c) Before redundancy letters are issued, the Superintendent of Education or designate will review with the President of the Local Bargaining Unit and one other member of the executive, the administration of Article 11.01.
- d) Where there is to be an exchange of data that will be required for any meeting between the parties, the parties will meet for the express purpose of exchanging such data if requested by either party. Required explanations of the data will be given at this information session.

11.02 - Surplus to the School

- (1) For the purpose of the staffing process, itinerant staff will be assigned 100% in the school where the majority of their teaching assignment occurs during the current school year.
- (2) In the event that a teacher's assignment during the current school year is equal between schools the teacher's school will be determined by lot.
- (3) A teacher surplus to the school shall be a teacher for whom no position (in accordance with Article 9.01 will be available in the present school in the upcoming school year. Where possible surplus will be declared based on the lesser of a teacher's contract time, 50% or 100%.
- (4) The Principal and Superintendent of Education will designate surplus teachers who shall thereafter be available for transfer (A.F.T.) by April 30.
- (5) Seniority with the Board shall be the determining factor where, in declaring a teacher surplus to the school, skill professional training and qualifications are deemed to be satisfactory subject to the right of the Board to retain sufficient qualified teachers to meet program requirements.
- (6) a) Before letters are issued declaring teachers surplus to the school, the Superintendent of Education or designate will review with the President of the Local Bargaining Unit and one other member of the executive, the Administration of Article 11.02.
 - b) Where there is to be an exchange of data that will be required for any meeting between the parties, the parties will meet for the express purpose of exchanging such data if requested by either party. Required explanations of the data will be given at this information session
- (7) Teachers employed on a permanent status who are declared surplus to the school will be offered, prior to May 15, the position of the least senior teacher in the system who has a timetable for the upcoming school year which the surplus teacher is qualified to assume or will become so qualified at the time the appointment becomes effective. The Board will accommodate the senior AFT teacher in accordance with the following modalities:
 - i) The parties recognize the responsibility of the Board, as provided for in the Education Act and Regulations, to provide the best possible program for its students.

- ii) Inherent in providing the best possible program is the requirement to have courses for students taught by teachers who are qualified to do so.
- iii) It is recognized by the parties that there are exceptions to the strict application of the requirements of the Education Act and its Regulations with regard to the definition of teacher qualifications in specific subject areas. As a result, the following procedure shall apply at all times.
 - (a) Teachers who instruct in a course at the Intermediate level which is not specifically designated in the Regulation as requiring OCT (Ontario College of Teachers) credits are deemed to be qualified.
 - (b) Notwithstanding i)(a) above, in the following subject areas, in order to be assigned a timetable, teachers must possess the required qualifications or obtain the needed qualifications by the date the assignment is scheduled to begin:
 - Intermediate and Senior Science
 - Intermediate and Senior Mathematics
 - Intermediate and Senior Computer Studies
 - Intermediate and Senior French Immersion (including the Board competency test).
 - (c) Prior to declaring a teacher surplus/redundant, the Board and the Branch Affiliate shall meet to determine whether the situation can be mitigated in favour of a senior teacher by the restructuring of proposed timetables. If such is the case, the Parties will undertake to effect such timetable restructuring as is necessary to ensure that the surplus/redundant Teacher(s) is/are the most junior teacher(s) to the school and the system.
 - (d) The Board will make its best and every effort to offer to a teacher who is declared surplus/redundant a timetable in accordance with his/her seniority and qualifications, subject to Subsection (b).
 - (e) It is understood that there may not be sufficient available positions for all surplus teachers.
 - (f) A teacher who does not possess qualifications for at least 50% of the courses taught on the timetable, subject to section (a) herein, will commit to taking a course leading to the necessary qualifications when that course is next offered.

- (g) The Employer recognizes that there may be circumstances that prevent a teacher from taking the course referred to in (f) when it is first offered, in which case the course will be taken at the next available opportunity. Any dispute as to whether the circumstances referred to in this section are valid will be subject to arbitration under Article 7.06.
- (h) Where a teacher needs more than one course leading to the necessary qualifications, the teacher will commit to acquiring the necessary qualifications in a period of one academic term longer than the necessary course time, subject to (g) above. This provision will only apply in one instance for each subject area.
- (i) The modalities outlined in (f) (g) and (h) do not apply to the subjects outlined in (b) above.
- (j) For the purpose of the agreement, a teacher is deemed qualified in Social Science if qualified in one of the following areas: History, Geography, Individual in Society, People and Politics, World Religions and Law.
- (k) For the purpose of this agreement, a teacher who is qualified in Business Studies is deemed to be qualified in Economics.
 - i) When adjustments are made to reflect actual enrolment, the staffing assignments shall be based on subject specific qualifications.
 - The Branch Affiliate shall be able to send representatives to participate in meetings, dealing with postings and the staffing process at the school and Board levels. The Association's participation in these meetings shall include making recommendations with the understanding that the final decision shall remain with the school Principal or the Superintendent of Education or designate as the case may be. It is further understood that the Local Bargaining Unit will be informed of all such meetings and retains the sole right to name teacher representatives to such meetings. The Local Bargaining Unit shall inform the Board in writing of the name of the representative chosen to attend these meetings.
- (8) The teacher so displaced in Article 11.02 (7) shall be declared surplus. Surplus teachers who have not obtained a position by May 15 will be declared redundant

- subject to the right of recall.
- (9) Prior to effecting the displacement process outlined in 11.02 (7), the parties agree to meet and consider alternatives in order to minimize its impact.
- (10) Notwithstanding the redundancy declaration date in article 11.01(2) and the surplus declaration dates referred to in articles 11.02(4), (7) and (8), these timelines may be extended by mutual consent of the Board and the Local Bargaining Unit.

11.03 - Posting

- (1) All vacancies in secondary schools for the following school year will be posted periodically between April 15th and the 3rd working day in June. Applications for the posted positions will be received until the 4th working day following the date of the posting. Applications will be supported by evidence of required qualifications. Vacant positions remaining after June 3 and new vacancies occurring after June 3 will be posted prior to June 30 and for a period of 4 working days.
- (2) Vacancies in semestered schools will be posted periodically between December 1st and the third working day in January for vacancies in the second semestered. Applications for the posted positions will be received until the 4th working day following the date of the posting. Applications will be supported by evidence of required qualifications.
- (3) All first and newly created positions in secondary schools occurring at other times than those listed above during the present school year, will be posted in secondary schools as they arise, plus one subsequent domino posting.
- (4) Posted positions and/or resulting vacancies will be filled thereafter in the following order <u>based on seniority and qualifications</u>:
 - a) Secondary teacher(s) A.F.T. and teachers employed in the Secondary panel who are not redundant.
 - b) Secondary teachers declared redundant within the <u>Local Bargaining Unit</u>.
 - c) Teachers on the Secondary recall list (as per Article 11.07)
 - d) In the event that a posted position and/or resulting vacancy remains

vacant at the secondary level, where qualifications, abilities and skills are equal, preference shall be given to teachers teaching Day School Continuing Education credit courses who have applied to the posting.

- e) Teachers from outside the Secondary Local Bargaining Unit.
- (5) All postings and vacancies shall be forwarded to the Local Bargaining Unit.

11.04 - Voluntary Specific Site and Term Transfer Request

- (1) Upon written application prior to March 31st of any school year a teacher may request a specific site and term transfer for the following school year or a portion thereof (semester). Such requests shall not be for a period of longer than two (2) school years. The Board shall consider all requests made pursuant to this article in consultation with the Local Bargaining Unit and prior to initial surplus declarations as per Article 11.02 of this agreement. Such transfer shall only take effect with the express approval of the teacher making the application, the Board, and the Local Bargaining Unit. Furthermore it is understood that a teacher transferred pursuant to this Article retains the option of return to the originating school and position, within the same subject area and discipline, subject to the criteria of the staffing process.
 - a) In the case of a single semester transfer, written notice is given by the teacher to the Board and the Local Bargaining Unit not later than November 30th in the case of a second semester and not later than March 31st in the case of a first semester transfer.
 - b) In the case of a full one or two school year transfer, written notice is given by the teacher to the Board and the Local Bargaining Unit by March 31st of the year in which the applied for term finishes.
- (2) Notwithstanding the above the term of transfer to a specific site may be extended if it is agreed between the teacher affected, the Board and the Local Bargaining Unit.

11.05 - Transfer of Staff Members without their Expressed Request

- (1) No staff member shall be transferred from one school or position to another except for reasonable cause.
- (2) The transfer shall be discussed with the staff member concerned prior to the

proposed transfer and official notice shall be given in writing.

- (3) When such a transfer is due to a teacher being surplus in a school the Board agrees to make every effort to arrange a transfer which is mutually satisfactory.
- (4) Between May 24th and June 20th for semester I and between January 15th and January 31st for semester II of each year, should a position for the following school year become available in a school where a teacher had been declared surplus to the school, the surplus teacher will be offered this newly vacant teaching position subject to seniority and required skills, professional training and qualifications.
- (5) Should such a transfer not be agreeable to the staff member concerned he/she shall have the right to appeal the decision individually or through his/her authorized representatives to the Director of Education.

11.06 - Procedure for Staff Reduction

The following procedures will be used when a reduction of staff is necessary.

- (1) This order shall be used in reducing staff:
 - a) Normal attrition
 - b) Probationary teachers
 - c) Permanent teachers
 - i) Seniority in the Bargaining Unit will be the determining factor where in the reduction of permanent teachers, skill, professional training and qualifications are satisfactory.
- (2) Subject to the provisions of 11.03 (4), where the Board has terminated the employment of any certificated teacher(s) as a result of staff reduction, it shall, when making new appointments to the staff, first offer such available positions to those eligible teachers who were terminated for reason of staff reduction. These re-appointed teachers shall retain all conditions of tenure which would have prevailed had service not been terminated.

11.07 - Right of Recall

(1) Teachers declared redundant in accordance with Article 11.01 will be placed on a list, Secondary Recall, in the order of their placement on the Secondary

seniority list. A teacher on the Secondary recall list will be rehired in order of seniority provided that the teacher has the required qualifications to fill the position or will become qualified prior to the commencement of the teaching position. In the event that the Board plans to recall a teacher who is not the most senior teacher on the recall list, by reason of qualification or available timetable, the Board and the Branch Affiliate shall meet to determine whether the situation can be mitigated in favour of a senior teacher by restructuring of proposed timetables in a manner that maintains course availability. If such is the case the Parties undertake to effect such timetable restructuring as is necessary.

- (2) Subject to the provisions of article 11.03 (4), any teaching position which becomes available will be offered to teachers on the recall list by verbal contact on the condition that they provide the Superintendent of Education or designate with an address and telephone number where they can be reached if other than their regular address and telephone number. They will be given 24 hours to accept or reject the position. If a teacher cannot be contacted verbally, a registered letter will be forwarded to his/her last address known to the Board. The teacher will be expected to advise the Board of his/her acceptance of the position within ten (10) calendar days of the mailing of the registered letter.
- (3) a) Refusal to accept a position of at least equal time will result in the teacher maintaining his/her position on the recall list for recall purposes. The redundant teacher will be offered a choice of positions available for which he/she is qualified or will be so qualified at the time the appointment becomes effective.
 - b) If a teacher on the recall list accepts a part-time position because a full-time position is not available, that teacher shall be offered the first full-time position that becomes available for which the teacher is qualified, or will be so qualified at the time the appointment becomes effective.
- (4) "Continuous Employment" will not increase for the period of time a teacher is on the recall list. For the purposes of Article 8.01, a teacher recalled from the "recall list" shall maintain their effective start date as if they had not been declared redundant.
- (5) Once a teacher is rehired all seniority and benefits will be restored.
- (6) If a teacher claims to be unable to accept a position at the time of recall due to illness, the Board at its discretion shall have the right to have the teacher

examined by a doctor selected by the Board.

(7) Teachers shall be on the recall list for a period of five (5) years from the date of termination of employment under Article 11.06

11.08 - Appointments of Teachers to Positions of Responsibility and Allowances

The appointment of teachers to positions of responsibility is done at the discretion of the Board and in accordance with the Education Act. The parties agree that a specialist/honours specialist is not a requirement under the Education Act for the position of Program Leader, however, this does not preclude the Board from identifying a preference for a specialist/honors specialist in Program Leader postings.

Positions of Responsibility are defined as Program Leader, <u>Consultant</u> and Co-ordinator.

- (1) Program Leader Positions:
 - (a) All positions of responsibility for Organizational Units shall be posted in accordance with Article 11:03.
 - (b) Appointment of teachers to positions of Program Leader shall be for a period of one year. If a position becomes vacant during the year, it shall be posted and filled for the remainder of the term.
 - (c) Teachers appointed as Program Leader of an Organizational Unit in a specific school must be assigned a teaching timetable in that school in each semester of that school year.
 - (d) Notwithstanding paragraphs (a), (b), (c) however, the appointment of teachers as Program Leaders in charge of Organizational Units does not prevent them from being affected by the normal application of the provisions of Articles 11:01, 11:02, 11:07 (Redundancy, Surplus and Recall Provisions).
 - (e) Teachers who are employed as Program Leaders shall receive a responsibility allowance as follows:

Effective September 1, 2008 3.0% \$4,563.17 Effective September 1, 2009 3.0% \$4,700.07

Effective September 1, 2010 3.0% \$4,841.07 Effective September 1, 2011 3.0% \$4,986.30

(2) Co-ordinator/Consultants Positions:

- (a) All positions of <u>Co-ordinator/Consultant</u>, as defined in the Education Act and Regulations, shall be posted in accordance with Article 11:03. Should a Co-ordinator/Consultant position/posting span both the <u>Secondary and Elementary panels</u>, the posting shall be open to <u>OECTA</u> members from both panels. The successful candidate shall continue to be covered under the Collective Agreement of the Local Bargaining Unit to which they currently belong.
- (b) Appointment of teachers to positions of <u>Co-ordinator/Consultants</u> shall be for a period of one school year. In special circumstances the Unit President and the Board may agree to terms of less than one school year. The teacher shall return to the staff of the last school in which the teacher taught subject to provisions of Articles 11.01, 11.02 and 11.07 at the end of the appointment.
- (c) Notwithstanding paragraphs <u>a)</u> and <u>b)</u>, the appointment of teachers as <u>Co-ordinator/Consultants</u> does not prevent them from being affected by the normal application of the provisions of Articles 11:01, 11:02, 11:07 (Redundancy, Surplus and Recall Provisions).
- (d) All qualifications being equal, preference shall be given to applicants presently in the Local Bargaining Unit.
- (e) Teachers who are employed as <u>Co-ordinator/Consultants</u> shall receive an annual responsibility allowance of:

Effective September 1, 2008 3.0% \$5,150.00 Effective September 1, 2009 3.0% \$5,304.50 Effective September 1, 2010 3.0% \$5,463.64 Effective September 1, 2011 3.0% \$5,627.54

(3) <u>Teacher In Charge</u>

(a) The parties recognize that, from time to time, school administrators (Principal/Vice-Principal) may be absent temporarily from their duties. To accommodate these situations, a teacher(s) may be designated as a

"Teacher In Charge" at a school. It is understood that this clause does not necessarily mean that the Teacher in Charge has been assigned these duties on every occasion that the Principal or Vice-Principal is away from the school, but only on those occasions where the duties have been specifically assigned.

- (b) No teacher shall be assigned without his/her consent.
- (c) A Teacher in Charge will remain a member of the bargaining unit for the duration of the duties assigned and will retain all rights and privileges accorded under the terms of the Collective Agreement.
- (d) "Teacher in Charge" shall be compensated according to the following daily rate when specifically assigned: Effective September 1, 2008: \$70.00 per day; Effective September 1, 2009: \$72.10 per day; Effective September 1, 2010: \$74.26 per day; Effective September 1, 2011: \$76.49 per day. Where the Principal specifically assigns these duties for less than a half-day, the Teacher in Charge shall receive 50% of this amount.
- (e) The Teacher in Charge shall not participate in the evaluation or disciplining of teachers, including occasional teachers, or any other Board employee in the school.
- (f) The Teacher in Charge shall be provided with the emergency contact numbers of the Principal, Vice-Principal(s) and Superintendent. The Teacher in Charge shall be informed of the administrator that is available through his/her contact number while there is no administrator in the school.
- (g) The services of a Teacher in Charge may only be called upon when the school administrator(s) (Principal/Vice-Principal(s)) are absent from the school for a period of ten (10) consecutive days or less.
- (h) The Teacher in Charge may be replaced by an occasional teacher. The parties shall jointly develop a protocol to ensure that the students will not be unsupervised as a result of a teacher acting in the role of Teacher in Charge.
- (i) The Teacher in Charge must be informed immediately when there are no administrators in the school.

(j) Teachers in Charge shall receive at least one-half (1/2) day of in-service on a school day no later than September 30th.

11.09 - Teachers Employed by the Board

All employment criteria being equal, preference shall be given to teachers presently employed by the Board.

11.10 - Posting of Teaching Positions and Positions of Responsibility

- (1) Teaching positions and positions of responsibility that become available during the school year, will be posted at all secondary schools and forwarded to the Unit President, <u>and</u> Executive Members. Positions that become available during July and August will be posted at the Board Office and forwarded to the Unit President and Executive Members. Whenever possible, the Board agrees to post positions on the Board Website <u>and forward them to the bargaining unit members via Board email.</u>
- (2) Teachers currently in the employ of the Board may apply for these teaching positions that are posted.
- (3) There will be no external advertising for any position, unless all qualified teachers on the recall list have been offered the positions.

11.11 - Job Sharing

- (1) The Board agrees to a job sharing plan.
- (2) Applications for such sharing must be made by May 31st and will be subject to the approval of the Director of Education or designate.
- (3) The portion of the position which is vacated, will be posted and filled according to the provisions of Article 11.03.
- (4) Sick leave credits shall be pro-rated according to the instructional time of the applicant.
- (5) The date of return to full-time teaching shall be specified at the time of application to job-sharing and shall not be less than one school year.
- (6) Each teacher involved in job sharing shall accumulate seniority as if the teacher

- were teaching full-time for a complete year.
- (7) The two teachers shall count as one teacher for the purpose of staffing.
- (8) The Board's contribution for teachers enrolled in the Board's Group Benefit Plan will be pro-rated according to the percentage of teaching time.

Article XII - Leaves of Absence

12.01 - Sabbatical Leave

- (1) The Board wishes to recognize members of its teaching staff who are giving outstanding service and offer them an opportunity for enrichment which in turn will benefit the school system. The Board reserves the right to determine if such leave is in the best interest of the school system and the decision of the Board in this regard is final.
- (2) A Sabbatical Leave may be granted for the following purposes: Education study, research or enrichment.
- (3) The duration of the Sabbatical Leave shall not be for more than one school year and shall not involve more than one academic year.
- (4) In order to qualify for Sabbatical Leave, a teacher shall have been employed by the Board in continuous service for not less than seven (7) years.
- (5) Selections will be made on the basis of the value of the Sabbatical Leave to the Board.
- (6) Application for Sabbatical Leave must be by written request (forms available) and be forwarded to the Director of Education prior to January 15th of any year to receive consideration. A request for Sabbatical Leave must have the approval of the Director of Education and such leave must be granted by resolution of the Board.
- (7) A teacher who has already been granted a Sabbatical Leave may apply for another Sabbatical Leave after a seven year period.
- (8) a) A teacher who accepts a Sabbatical Leave must remain with the Board for three (3) years after returning from leave. Such a commitment shall be in writing.

- b) A teacher who does not remain in the employ of the Board for three (3) years after returning from the leave shall pay to the Board 33% of the salary received during the leave for each year fewer than three that he/she does not remain in the employ of the Board.
- c) A teacher who has benefitted from Sabbatical Leave and suffers a disability or illness which prevents him/her from performing his/her duties, will be expected to fulfill the conditions of this agreement as set out in the above paragraph upon resumption of his/her duties.
- (9) A teacher on Sabbatical Leave shall receive the following benefits:
 - a) Seventy percent exclusive of salary allowances for additional responsibilities. A maximum of an additional thirty percent exclusive of salary allowances for additional responsibilities may be withdrawn from the cumulative sick leave plan.
 - b) All salary adjustments, increments and medical benefits.
 - c) Unless changes have occurred which would alter the teacher's ability or personal commitment to assume the responsibilities he/she had prior to his/her Sabbatical Leave, a teacher upon his/her return will be subject to Article 12.11.

12.02 - Leave for Personal Reasons

- (1) When teachers are required to be absent for personal reasons, they shall be granted up to two (2) complete days or four (4) half-days per school year, taken individually, collectively, or in combination, without loss of pay, but subject to deduction from sick leave credit.
- (2) Personal Leave days shall not be taken on a <u>professional development day or on</u> the school day immediately preceding and/or following a holiday or a vacation period unless the leave is approved by the Director of Education or designate.
- (3) The Teacher shall make every reasonable effort to notify the Principal of the date(s) and length of the Personal Leave day(s) at least five (5) days in advance, except in the case of an emergency. In the case where the Teacher is not employed at a school, he/she shall notify the immediate supervisor.

12.03 - Leave of Absence due to Lengthy Illness

- (1) A teacher who becomes seriously ill and is forced to use all of his/her current sick leave and sick leave credit reserve, may request a leave of absence from the Board prior to the expiration of his/her sick leave credits.
- (2) A teacher who applies for Long Term Disability Benefits shall be deemed to be on a leave of absence for Lengthy Illness as of the date upon which he/she becomes entitled to receive such benefits.
- (3) If such a leave is requested by the teacher, the Board shall grant a leave of absence for either:
 - a) the remainder of the school year in which the leave was requested, or
 - b) the remainder of the school year, plus the following three (3) school years, or
 - c) any other length of time that is mutually agreeable to both the Board and the Teacher.
- (4) Upon the expiration of the leave, a teacher must advise the Board in writing of his/her ability to resume his/her teaching duties.
- (5) In all cases, the Board at its discretion shall have the right to have the teacher examined by a doctor selected by the Board.
- (6) Sick leave credits and experience will not accumulate during the time spent on this leave.
- (7) A member on LTD leave retains his/her employment status with the Board throughout the period of the leave as defined in article 12.03(3), and maintains the right of a member under the terms of the Collective Agreement.
- (8) A member on LTD leave of absence or absence due to lengthy illness shall continue to accumulate seniority at the same rate as if he/she were not on leave.
- (9) Any member on LTD leave of absence or absence due to lengthy illness shall be subject to the provisions of Article XI during the period of the leave.

12.04 - Leave of Absence (One year, one semester, part-time leave of absence)

(1) a) A leave of absence without pay <u>for one year</u> may be granted by the Board

to teachers who have submitted a written request to the Director of Education prior to March 31 of the year in which the leave is to commence.

- b) A leave of absence without pay for one semester may be granted by the Board to teachers who have submitted a written request to the Director of Education prior to November 30th if the leave is to commence Semester II or prior to March 31st if the leave is to commence Semester I of the following year.
- c) i) A teacher with a full-time assignment who, prior to March 31st, requests a part-time leave commencing the following school year may have the leave request granted by the Director of Education or designate.
 - ii) A teacher with a full-time assignment who, prior to March 31st, requests a part-time leave for the first semester commencing the following school year or who, prior to November 30th, requests a part-time leave for the second semester of the current school year may have the leave request granted by the Director of Education or designate.
 - iii) A teacher who requests and is granted a part-time leave for a specified period will return to a full-time assignment at the end of the leave, subject to Article XI.
 - iv) A teacher may apply prior to the <u>deadlines in i) and ii) above</u> for an extension to the teachers part-time leave and such extension will be considered subject to staffing requirements.
- d) Changes in the terms of a leave of absence under this Article may be made only by the mutual consent of the teacher, association and the Board. Changes must be in writing and conform with the requirement of this Collective Agreement.
- (2) Leaves of absence shall be granted to teachers on the basis of their seniority with the Board and the date of their request.
- (3) a) Written acceptance or denial of the teacher's request for a one year leave of absence or one year part-time leave of absence with explanation, will be forwarded to the teacher by April 30th in the school year the original

request is made.

- b) Written acceptance or denial of the teacher's request for a semester leave or a one semester part-time leave of absence, with explanation, will be forwarded to the teacher by <u>December 15th</u> for a semester II leave or by <u>April 30th</u> for a semester I leave.
- (4) A teacher returning from a one year, <u>one semester or part-time</u> leave of absence will be subject to article 12.11.
- (5) Sick leave credits and experience will not accumulate during the time spent on this leave.
- (6) Once a request is granted, the teacher will be obligated to take his/her leave of absence <u>unless mutually agreed upon by the teacher, the Board and the local bargaining unit.</u>
- (7) Notwithstanding the application deadlines in this article, applications received beyond the above stated deadlines, may be considered by mutual consent of the Board and the local bargaining unit.
- (8) Any leave approved under this article shall be reflected by an equivalent adjustment in the FTE positions relevant to Articles 11.01, 11.02, 11.03, except for leaves approved after September 1 for semester II.

12.05 - Leave of Absence for <u>Unit</u> President <u>and/or Unit Release Officer(s)</u> of OECTA

- (1) The local <u>bargaining unit</u> will have the option of selecting one of the following plans:
 - a) Upon giving reasonable advance notice to the Director or Education, a maximum leave of absence of ten (10) days during the school year with the replacement cost of the teacher on leave to be reimbursed by the Local Bargaining Unit at the current occasional teacher daily rate; or
 - b) a percentage of release/leave time for the unit president and/or unit release officer(s) as determined by the local bargaining unit.
 - c) Regarding the release/leave time allocation for the Unit President's and/or unit release officer(s)'s leave in b), the teaching assignment of the

president <u>and/or unit release officer(s)</u> shall be determined in consultation with the Superintendent and the Unit President and will be subject to the staffing process. The President <u>and/or unit release officer(s)</u> shall accumulate full seniority during the duration of the leave.

- (2) When the local bargaining unit selects plan (b) above, it will notify the Director of Education or designate of its percentage of total release/leave time and the allocation thereof to the Unit President's and/or unit release officer(s)'s release prior to April 1st in the year of the leave in order to qualify for the plan.
- (3) The total cost of the salary and benefits of the <u>Unit President and/or unit release</u> officer(s) on leave under paragraph (1) (b), plus any additional President's allowance approved by the Bargaining Unit of the Teacher on Leave will be reimbursed in full to the Board by the Local Bargaining Unit.
- (4) In the event that the <u>Unit President or unit release officer(s)</u> is unable to perform <u>their duties</u> for a period of more than twenty (20) consecutive school days due to injury or illness, the Unit may appoint another teacher as Interim President or unit release officer until the President <u>or unit release officer</u> is able to return to his/her duties.
- (5) The Board shall grant a leave to the teacher named by the Association as Interim President or unit release officer during the period of the appointment noted in 12.05 (4).
- (6) A teacher returning from an Association leave has a right to be reassigned to the same school as the teacher last taught in and in the same subject area, subject to Article XI.

12.06 - Deferred Salary Leave Plan (formerly X/Y Leave of Absence)

(1) Description

This leave plan has been developed to afford teachers the opportunity of taking a leave of absence with pay for one year or one semester.

(2) Application

a) A teacher must make written application to the Director of Education on or before February 28th for plans beginning in September, and on or before December 1st for plans beginning semester II requesting

permission to participate in the Plan.

- b) Written acceptance or denial of the teacher's request, with explanation, will be forwarded to the teacher by May 25th for plans beginning in September and by January 10th for plans beginning semester II..
- c) If applications permit, the Board will accept a maximum of 3% of their present teaching staff as of September 1st, of the current school year. Leaves will be granted on the basis of seniority.

(3) Payment Formula

- a) While on the Deferred Salary Leave Plan the salary earned during the work period will be evenly distributed over the plan period in accordance with 5.02(1) Method of Payment.
- b) The sums deducted in accordance with (a) above shall be placed in an account and bank or other financial institution of the teacher's choice; such account to be in the Board's name identified to the individual teacher's name. All additional sums generated by this account will be paid on the last instalment date.
- c) During the plan period, employee benefits shall be maintained in accordance with Article VI. The Board shall pay the premiums prorated to the percentage of the work period. The Employee shall pay the premiums prorated to the percentage of the leave period.
- d) The teacher shall continue to receive full sick leave credits and experience during the work period of the plan.

(4) Position on Completion of the Leave

- a) On return from leave, a teacher will be subject to Article 12.11.
- b) Sick leave credits and experience will not accumulate during the year spent on leave.
- c) No one will be granted a leave under this Plan who has been on a Sabbatical Leave and has not fulfilled all of the requirements of their previous leave.

- d) Teachers declared redundant while participating in the Plan will be required to withdraw and will be paid a lump sum adjustment for any monies deferred to the date of withdrawal, plus any interest earned in the account described in 4 (b). Repayment shall be made within sixty (60) days of the withdrawal from the Plan.
- e) Pension deductions are to be continued as provided by the Teacher's Pension Plan Act.
- f) A teacher may withdraw from the Plan any time prior to February 28th for leaves commencing in September and prior to November 15th st for leaves commencing semester II or a mutually agreed upon date. Upon withdrawal, any monies plus interest accumulated in the account described in 3 (b) will be repaid to the teacher within sixty (60) days of notification of his/her desire to leave the Plan.
- g) In the event that a suitable replacement cannot be found for a teacher who has been granted a leave, the Board may defer the leave for a mutually agreed upon time period by so advising the teacher prior to February 28th for leaves commencing in September and prior to November 15th for leaves commencing semester II or a mutually agreed upon date. In this instance, a teacher may choose to remain in the Plan or may withdraw and receive any monies and interest accumulated in the account, described in 3 (b), to the date of withdrawal. In the latter case, payment shall be made within sixty (60) days of the date of withdrawal.
- h) A teacher in this plan may defer the leave for a mutually agreed upon time period by so advising the board prior to February 28th for leaves commencing in September, and prior to November 15th for leaves commencing semester II or a mutually agreed upon date.
- i) Should a deferral result in a leave being taken past the leave period of the Plan, any monies accumulated by the termination date of the Plan, will continue to accumulate interest until the leave is granted.
- j) All teachers wishing to participate in the Plan shall be required to sign the necessary form(s) supplied by the Board before final approval for participation is granted.
- <u>k)</u> Should a teacher die while participating in the plan, any monies plus interest accumulated, in the savings account mentioned in Article 12.06

- c) ii), at the time of death, will be paid to the teacher's estate.
- l) <u>In accordance with Canada Revenue Agency guidelines the teacher must return to employment with the Board for the same duration as the leave period.</u>

12.07 - Pregnancy/Parental Leave (Including Adoption)

- (1) Pregnancy/Parental/Adoption Leave shall be granted to a teacher in accordance with the Employment Standards Act as revised and pertinent provincial Legislation as amended from time to time.
- (2) The Board shall pay 95% of the employee's salary for the first two (2) weeks of Employment Insurance based on 260 days as per Employment Insurance Program.
 - In addition, teachers have the option of choosing one of either 2 (a) or 2 (b) below. Once the decision has been made, it cannot be reversed for the duration of the leave:
 - a) The Board shall pay the difference between Employment Insurance payment and 75% of the employee's salary for the next eight (8) weeks based on 260 days as per Employment Insurance Program. These payments constitute a modified Supplemental Employment Benefit payments plan (SEB), which shall be subject to the approval of Human Resources and Social Development Canada. To receive pay, the employee must provide proof that she/he has applied for and is in receipt of Employment Insurance maternity or parental benefits.

OR

b) The teacher, who is the natural birth mother, may utilize her sick leave credit up to a maximum of six (6) weeks following the birth of his/her child, after which the Board shall pay the difference between Employment Insurance payment and 75% of the employee's salary for the next two (2) weeks based on 260 days as per Employment Insurance Program. These payments constitute a modified Supplemental Employment Benefit payments plan (SEB), which shall be subject to the approval of Human Resources and Social Development Canada. To receive pay, the employee must provide proof that she has applied for and is in receipt of Employment Insurance maternity or parental benefits, and must provide medical proof confirming the actual date of delivery.

Participation in SEB plan shall not allow the teacher to be eligible for more than 100% of the teacher's salary and applicable allowances.

Notwithstanding the above, if serious medical complications beyond that of a normal birth of a child occur or other serious medical issues arise, the teacher may access her available sick leave.

- (3) Extended Parental Leave without pay may be granted to a teacher on staff up to a maximum of two school years. The teacher shall return from the extended parental leave on a date mutually acceptable to the Board and the teacher.
- (4) The terms of Article 12.07 shall apply immediately upon a teacher's employment.
- (5) The Board shall maintain its share of the premiums for all benefits described under Article VI during the statutory period of a teacher's pregnancy/parental/adoption leave as per Article 12.07.
- (6) Receipt of paid sick leave shall not affect a teacher's entitlement to SEB benefits pursuant to the Pregnancy/Parental/Adoption Leave Article for those portions of her pregnancy leave for which she is not in receipt of paid sick leave.
- (7) Early Return A teacher may terminate a pregnancy/parental/adoption leave and return to work upon providing the Board with two (2) weeks written notice.
- (8) Child Care Leave for Medical Reasons Upon request a teacher shall be given an unpaid child care leave for medically approved reasons for the child for up to two (2) years. The Board at its sole discretion may approve an extension of such leave up to the child's third (3rd) birthday.

12.08 - Special Time Off

All teachers shall be permitted:

- (1) Without deduction from salary, benefits, or sick leave credit.
 - a) Special time off to write examinations for the improvement of professional qualifications or one (1) day off to attend own graduation.
 - b) Special time off if he/she is charged with a criminal or quasi-criminal offence alleged to have occurred while performing School Board duties.

Such teacher shall receive his/her full salary for the days of absence from school while attending trial if he/she is acquitted of such offence or the charge is withdrawn. Salary for days of absence at trial will not be paid if the teacher is convicted.

c) Paternity Leave:

Special time off up to one day shall be allowed to a teacher upon the birth of a child.

d) Adoption Leave:

Special time off up to one day shall be allowed to a teacher upon the adoption of a child.

- e) Juror or Witness Duty: Special time off to serve as a juror or witness in any proceeding.
- f) Special time off to be granted to an employee for a period of quarantine when declared by the Medical Officer of Health or designate.
- (2) Special time off without salary deduction shall be granted for attendance at provincial executive or committee meetings of OECTA or OTF or as representatives of same at educational functions. The local bargaining unit shall reimburse the Board at the current occasional teacher daily rate per teacher if a supply teacher is called in. No single teacher may request or be granted more than fifteen (15) days in one academic year.
- (3) Special time off with salary deduction shall be permitted to a teacher who is required to appear before a court or tribunal as a party to the litigation.
- (4) Special time off without salary deductions but with deductions from sick leave for exceptional circumstances such as a serious illness in the immediate family, for a period of up to three (3) days per year, may be granted upon the approval of the Director of Education or designate. Part-time teachers' entitlement and usage shall be pro-rated accordingly.
- (5) Special time off to a maximum of ten (10) days without salary deduction shall be permitted for a teacher who is required to fulfill the responsibilities of Chief Negotiator. This leave shall be permitted subsequent to the commencement of negotiations. The Bargaining Unit shall reimburse the board the sum of the current supply rate per diem per teacher if a supply teacher is called in.

12.09 - Bereavement Leave

Each eligible teacher shall be allowed leave of absence without deduction of salary and without deduction from sick leave credits as follows provided notification is given to the Superintendent of Education or designate:

(1) Up to a maximum of five (5) consecutive working days upon the death of a:

spouse	brother	step-child	grand-parent
child	sister	step-parent	step-brother
parent	guardian	grand-child	step-sister

(2) Up to a maximum of three (3) consecutive working days upon the death of a:

parent-in-law	brother-in-law	spouse's brother-in-law
son-in-law	sister-in-law	spouse's sister-in-law
daughter-in-law	grand-parent-in-law	

(3) Up to a maximum of two (2) working days in any one school year, to be used separately or together, upon the death of:

an aunt	a spouse's aunt	a close friend
an uncle	a spouse's uncle herein	any other family member not listed

- (4) It is understood that the leaves under 12.09 (1), (2) and (3) above may not be taken if one week or more of a vacation period remains at the time of death.
- (5) one day maximum to attend the burial of a relative covered under paragraph (1) of this clause if the death has occurred during the winter months.
- (6) an additional day or days without deduction of salary may be granted for special circumstances such as excessive travelling as approved by the Director of Education or his/her designate.

12.10 - Special Leaves of Absence

A leave for reasons not otherwise addressed in this agreement may be granted at the discretion of the Director of Education to a teacher who has so requested in writing. Such leave will be without pay, with appropriate proration of accumulation of sick leave and the approval or denial of same will not be grievable.

12.11 - Return from Leave

Teachers returning from any approved leave shall be given a comparable position in the same school where such a position is available in accordance with Article XI.

12.12 - Special Leave Incentive Plan

This article applies only in situations of redundancy as of April 15th for semester 1 and December 15th for semester 2, where a recall list exists and the replacement teacher can be taken from that recall list. These terms apply equally in situations where the leave is requested for either or both semesters in the same school year.

- (1) A teacher taking a half-time leave under this plan will be paid 10% of base salary on the second scheduled pay of September for a Semester I leave and February for Semester II leaves while on leave, that is, 60% of his/her annual salary that school year. The Board will pay 100% of the benefits of a full-time employee while on this leave. Benefits for part-time employees will be pro-rated.
- (2) Teachers wishing to take advantage of this leave plan are to apply in writing to the Director of Education designating the semester being chosen for the leave. Applicants are also requested to state whether a leave for other semesters would be acceptable, if the original request cannot be met. Letters of application must be received by the Director of Education by March 31, or the last day before March 31.
- (3) Only secondary school teachers with twelve (12) or more years of teaching service are eligible for this plan.
- (4) The Board reserves the right to limit the number of leave totals and the number of leaves per semester, to ensure viability of program.
- (5) A leave under this plan will only be approved if ultimately the overall cost of salary plus benefits of the replacement teacher plus the cost of the teacher on leave does not exceed the cost that the Board would have incurred had the teacher not taken the leave of absence.
- (6) Teachers will be notified in writing whether or not the leave has been approved on or prior to June 20 if the leave is for semester I and the latter of October 15th or upon completion of the September 30th staffing adjustments, if the leave is for semester II providing that teachers agree to waive the obligation under 11.03 (2)

for second semester vacancies due to an absence approved under 12.12.

- (7) Teachers returning from a leave under this Plan shall return to the same school and, if applicable, to the same position of responsibility.
- (8) A teacher from the recall list hired as the replacement teacher maintains his/her right of recall to a permanent teaching position.

12.13 - Family Medical Leave

Family Medical leave shall be granted to a Teacher in accordance with the Employment Standards Act.

Teachers returning from Family Medical leave shall be entitled to the same position held prior to the leave, subject to the staffing process.

Article XIII - Teaching Conditions

13.01 - Teaching Load

- (1) The school Principal shall make every effort:
 - (a) not to schedule a teacher for (3) three consecutive teaching periods when teaching (3) three classes in a semester;
 - (b) to minimize the number of preparations required by a teacher.
- (2) Where practical and possible, teaching assignments for the upcoming school year will be made available before the end of the current school year. If changes are made during the summer, every effort will be made to notify the teacher of the change.
- (3) The teaching workload of each full-time teacher shall be 3 periods out of 4 periods each semester and these three periods will be scheduled during the regular instructional day. All teachers will have an unscheduled fourth period.
- (4) Notwithstanding the previous paragraph (3), a principal and/or the Board may request an exception for an alternate day (before and/or after school) program

credit course and for a teacher workload imbalance for the delivery of this program credit course (e.g. Concert Band). Such exceptions must be agreed upon yearly no later than May 1st for the upcoming school year by all of the following parties: the school principal, the Superintendent of Schools, the teacher affected and the Local Bargaining Unit. A schedule of instruction will be agreed upon to ensure viability of the course and credit, and a modified supervision/on-call schedule will be agreed upon for the teacher.

- (5) The target number of students per section in the Co-operative Education Program will be eleven (11) students.
- (6) If guidance teachers, Student Success Teachers or other teachers are requested by the principal or Board designate to work after the last school day in June and before the first school day in September, they shall do so on a voluntary basis and will not be compensated with equivalent time off during the school year.

13.02 - Personnel Files

- (1) A teacher shall have access during normal business hours to his/her personnel files, whether in school or at the Board Office. The teacher may copy any material contained in these files.
- (2) If the teacher disputes the accuracy or completeness of any such information, the Board shall, within 15 days from receipt of a written request by the teacher stating the alleged inaccuracy, either confirm or amend the information.
- (3) Where the Board amends information, the Board shall, at the request of the teacher, notify all persons who received a report based on the inaccurate information.
- (4) Where a teacher gives written authorization to another person acting on the teacher's behalf, the Board shall provide such access, as well as copies of materials contained therein, if also authorized and requested.
- (5) When any additions, deletions or modifications are made to a teacher's personnel file, the Board shall provide copies of any such additions, deletions or modifications to the teacher within five (5) working days of such changes.

13.03 - Performance Appraisal

(1) The Performance Appraisal of Experienced Teachers (PAET) and the

Performance Appraisal and New Teacher Induction Program (NTIP), as defined by the legislation, shall be conducted in accordance with the Education Act, regulations 98/02, 99/02, and the Ministry of Education's PAET Technical Requirements Manual, 2007, Manual for Performance Appraisal of New Teachers, 2006, and Induction Elements Manual, 2006 and any of which may be amended from time to time. No additional domains, competencies and look-fors beyond those outlined in the Teacher Performance Appraisal document developed by the Institute for Catholic Education shall be used in the performance appraisal of a teacher.

- (2) No member of the bargaining unit shall participate in the evaluation of another employee of the Board.
- (3) "Performance Appraisal" shall mean an assessment of a classroom teacher's performance by a Supervisory Officer of the Board and/or Principal, which will result in a written summative report which will be placed in a teacher's personnel file.
- (4) A pre- <u>observation meeting will be held</u> with the teacher to mutually discuss <u>the focus of the classroom observation</u>. This meeting will be held at least two (2) instructional days prior to the day of the <u>classroom observation</u>.
- (5) The <u>classroom observation</u> shall be <u>conducted</u> only after five (5) days' notice to the teacher involved. A post <u>observation meeting will</u> be held with the teacher to discuss the <u>performance appraisal</u>. This post <u>observation meeting</u> will be held within five (5) instructional days after the <u>classroom observation</u>.
- (6) The summative report shall be made available to the teacher at the earliest possible opportunity, but no later than twenty (20) instructional days from the date of the classroom observation. At the request of either the teacher or the principal, the parties shall meet within five (5) instructional days of the summative report to discuss the performance appraisal. A union representative can accompany the teacher and the principal may also invite a second party. Once completed, the appraisal form will be sent to the Superintendent for review and then placed in the teachers' personal file.
- (7) The Board recognizes that extracurricular activities are voluntary. Unless mutually agreed by the teacher being appraised and by the principal, vice-principal or supervisory officer conducting the performance appraisal, voluntary activities shall not be a part of the performance appraisal process.

- (8) The Board will provide a list of teachers in the performance appraisal year to the President of the Secondary Local Bargaining Unit by October 31st of each school year.
- (9) The Board shall notify the President of the Secondary Local Bargaining Unit of any teachers receiving a performance appraisal rating other than a "satisfactory" within five (5) days.
- (10) It is understood that a teacher's Annual Learning Plan (ALP) is an individual professional growth document. Consultation and collaboration related to the ALP shall take place in the method as defined by the Ministry of Education.
- (11) The parties agree to abide by the Memorandum of Agreement on Mentoring Guidelines, NTIP of September 2006 or any mutually agreed upon revisions.

13.04 - Offence Declarations

Every teacher is required to complete an Offence Declaration annually by September 1. The Declaration shall be <u>treated in a confidential manner and</u> accessible only to the Director or a designate, <u>the Superintendent of Employee Relations or a designate</u>, <u>the Human Resources Manager or a designate</u> and the individual teacher who is referenced.

13.05 - Credit Rescue and Credit Recovery

- (1) Upon the end of the semester, a teacher's responsibilities regarding credit rescue and/or recovery ends with the completion of the failure form and credit recovery recommendations.
- (2) It is the responsibility of the Credit Recovery teacher to mark and/or evaluate any student work submitted after the end of the term for consideration in a final report card.
- (3) Notwithstanding (1) and (2) above, a teacher may volunteer, at their own discretion, to mark or evaluate any student work submitted as part of a credit rescue and/or recovery program/credit.

13.06 - Joint Professional Development Committee

The Board and the Secondary Local Bargaining Unit are committed to the continuous development of a Catholic Professional Development Community in each of the

schools of the Board and system-wide, and, to that end are committed to fostering an atmosphere within each of the schools and system-wide that promotes a focus on learning, collegiality, respect for professionalism, continuous learning, collective inquiry into best practices, innovation and experimentation, all in order to improve teaching and student learning.

The Board and the Local Bargaining Unit agree that professional development is job-embedded, and formed by research, done in partnership with colleagues and is to be informed by the Teachers' Annual Learning Plans.

- (1) The Board agrees that Board-initiated staff development and in-service that Teachers are required to attend shall be provided during the instructional day and, where reasonable and practical, with a minimum of five (5) days notice.
- (2) Within ninety (90) days of ratification of this collective agreement, there shall be established a Joint Professional Development Committee (the "PD Committee") composed of three (3) representatives appointed by the Board and three (3) representatives appointed by the Secondary Local Bargaining and mutually agreed appropriate resource staff.
- (3) The designated representative of each party shall chair meetings alternately. The PD Committee shall be convened no later than September 30, in each school year, for an initial meeting. The committee will draft its own terms of reference, which will include content and frequency of meetings. The parties shall meet at least three (3) times per year.

(4) The PD Committee shall:

- a) Address ways in which funds generated by the allocation in the Grants for Student Needs to enhance professional learning opportunities for teachers will be used.
- b) Ensure that professional activities for teachers during Professional Activity Days are consistent with the learning goals identified in the Teachers' Annual Learning Plans.
- c) Promote best practices in sustaining successful Catholic Professional Learning Communities and in monitoring their implementation in the schools of the Board and system-wide.
- d) The Committee will provide advice and assistance to Board staff that are assigned responsibility for providing professional development to teachers and planning for

such activities.

13.07 - Faculty of Education Student Placement

The Board and the Local Bargaining Unit believe that Faculty of Education student candidate placements form an integral part of a future teacher's development and recognize the volunteer participation of teachers in accepting, mentoring and supervising Faculty of Education teacher candidates.

13.08 - Credit Integrity

In the event that an administrator with the authority to change marks, as outlined in the Education Act, chooses to exercise their right, the administrator will notify the teacher of such change prior to the reporting, where reasonable and practical.

Article XIV - Supervision

14.01 - <u>Uninterrupted Lunch</u>

All teachers shall have an uninterrupted forty (40) minute lunch period.

14.02 - On call Periods, Supervision Periods and Preparation Time

- (1) Each full-time teacher will be assigned 6 periods out of 8 periods. Each full-time teacher may also be assigned up to the following maximum Additional Professional Assignments comprised of on-calls/supervisions.
 - a) effective September 2008 38 half periods of on-calls/supervision
 - b) <u>effective September 2009 36 half periods of on-calls/supervision</u>
 - c) effective September 2010 33 half periods of on-calls/supervision
 - d) effective September 2011 30 half periods of on-calls/supervision

Part time teaching assignments will be prorated.

- (2) Teachers who have schedules that require travel between campuses and/or schools during the lunch period shall be exempt from supervision as intended in Article 14.02(1) above.
- (3) A teacher shall be available for Home Room in the fifteen (15) minutes prior to the commencement of the first instructional period of the day. Such time shall not constitute supervision/on-call or instructional time. Any assigned supervision

during the times as outlined above, such as but not limited to, bus duty, hall duty and/or yard duty, shall constitute supervision/on-call time for the purpose of the minutes of supervision/on-call maxima as set out in 14.02 (1).

- (4) Teachers will be available for other forms of supervision only in emergency situations, that is where there are no other personnel reasonably available for supervision.
- (5) No teacher shall be assigned a half-period on-call/supervision and lunch supervision on the same day.
- (6) The frequency and number of minutes of supervision shall be assigned in a fair and equitable manner at each school.
- (7) <u>Each teacher's supervisions/on-calls shall be recorded by the school and made available to the teacher for review upon request.</u>

Article XV - Other Matters

15.01 - Association Fee

The Board shall deduct from the pay of each teacher who is within the scope of this Agreement, ten (10) equal installments for the fees established by the Association. The Association shall advise the Board in writing of the amount of the fees authorized by the Association membership in keeping with the Constitution and By-laws of the Association. The Board shall remit the total amounts so deducted to the Ontario English Catholic Teachers' Association within twenty (20) working days of collection.

15.02 - Local Levy

The Board shall deduct from each Teacher's pay in September the regular levy dues of the Local Bargaining Unit of the Ontario English Catholic Teachers' Association.

The Local Bargaining Unit shall advise the Board in writing of the amount of the dues authorized by the membership in keeping with the constitution and by-laws of the Local Bargaining Unit. The Board shall transmit the total amounts so deducted to the Treasurer of the Local Bargaining Unit of the Ontario English Catholic Teachers' Association.

15.03 - Printing and Distribution of this Agreement

- (1) The agreement shall be printed in booklet form and the printing cost shall be shared equally each year by the Board and the teachers.
- (2) Each teacher in the employ of the Board shall receive a copy of this agreement as soon as possible after the official signing of the document.
- (3) Any teacher to be hired by the Board shall be provided with a copy of the current agreement and a copy of a letter substantially in the form of the Compensation and Health Benefits Attachments in Appendix 1, along with the information specified in it, within the first week of employment.

15.04 - Just Cause

- (1) No teacher shall be discharged or otherwise disciplined without just cause.
- (2) The Board shall provide the teacher with written notice of termination of employment.
- (3) Such notice shall state the reason(s) for termination.
- (4) Such notice shall be sent to the teacher's last known address by registered mail or via hand delivery.
- (5) Notwithstanding the above provisions, notice of termination of employment due to redundancy shall be in accordance with Article 11.01 of the collective agreement.

15.05 - Board Statement to Teachers

A teacher may receive any or all of the following information in a statement by providing a written request to Human Resources, clearly outlining the information required. Such request shall not be on a more frequent basis than once a year.

- 1) the amount of accumulated sick leave
- 2) the number of years experience
- 3) the Q.E.C.O. level placement
- 4) the grid component of their salary
- 5) the annual salary
- 6) seniority with the Board.

15.06 - Continuing Education

- (1) a) "Continuing Education Teacher" as referred to in this article shall mean an individual who falls within the term "Teacher" as defined in the Education Act, as amended, who is engaged in the teaching of a continuing education course.
 - b) A "Continuing Education Course" shall mean a course developed from the Ministry of Education guidelines or approved by the Ministry, which requires the person delivering the course to hold a valid College of Teachers Certificate in good standing and which is eligible for continuing education
 - c) i) All modules will be offered to teachers on the list in order of seniority with the Board.
 - ii) The salary and benefits for Continuing Education teachers will be as set out in this agreement.
 - iii) All Continuing Education teachers on the seniority list are deemed to have continuous employment with the Board, provided they continuously accept assignments for which they are qualified.
- (2) Summer School credit courses shall be delivered by Bargaining Unit members and posted as per Article 11.03(4).

15.07 - The Adult Continuing Education Program

- (1) For the purpose of this program, an adult is defined as a person who is 21 years of age or older. The Adult Continuing Education Credit Program provides adults with the opportunity to complete the requirements of an OSSD or to earn credits for personal or work related reasons.
- (2) A teacher(s) will be hired for the Program in accordance with The Act and regulations and as per Articles 8.04 and 8.05 subject to the right of the Board to retain sufficient qualified teachers to meet program requirements. All terms and conditions of this Collective Agreement that apply to Continuing Education teacher(s) will apply to teachers in the Program.
- (3) A teacher employed in the Program will be paid at the hourly rate of \$38.65 effective September 1, 2008, \$39.80 effective September 1, 2009, \$41.00

- effective September 1, 2010 and \$42.25 effective September 1, 2011, plus 4% vacation pay.
- (4) For greater certainty, teachers who teach credit courses to students under 21 years of age in the alternative program are not continuing education teachers, and are covered by all terms and conditions of this Collective Agreement.
- (5) Teaching staff at St. Albert Adult Learning Centre will be granted a total of five (5) sick days per year. This is non-cumulative.
- (6) A teacher employed in the program shall be entitled to bereavement leave of absence under Article 12.09 Bereavement Leave.
- (7) Any new or vacant posted Adult Continuing Education Program positions that become available will be filled in the following manner:
 - a) by seniority and qualification from the Day School Continuing education Seniority List established in Articles 8.04 and 8.05
 - b) qualified secondary teachers declared redundant or on the recall list within the Local Bargaining Unit
 - c) qualified teachers declared redundant within the Board
 - d) qualified teachers from outside the Board
- (8) In the event of a layoff or non-renewal of an Adult Continuing Education teacher contract, as per Articles 15.07, 15.08, 8.04 and 8.05, due to declining enrollment or the suspension/cancellation of the Adult Education Program, the Board shall place the affected teacher on the Occasional Teachers' "B List", subject to the Board's rights under Article 3.01.

15.08 - Reopener Clause

With the agreement of both parties, negotiations of any particular clause(s) may be reopened

during the life of this agreement.

15.09 - Letters of Understanding - Letters of Intent

Both parties agree that all letters of intent and letters of understanding contained in this agreement

are subject to the grievance procedure as a means to dispute an alleged non-compliance by one of the parties.

15.10 - Freedom from Assault in the Workplace

- (1) The Board and the Local Bargaining Unit recognize that every employee has a right to freedom from assault in the workplace. To this end, the Parties recognize that the Board will maintain and share with the Local Bargaining Unit a procedure for consistently dealing with instances of alleged staff assault by a student.
- (2) The Board and the Local Bargaining Unit recognize that every employee has a right to freedom from harassment in the workplace. To this end, the Parties recognize that the Board will maintain and share with the Local Bargaining Unit a policy for consistently dealing with instances of alleged harassment.

15.11 - Denominational Rights

Nothing in this collective agreement shall be construed in such a way as to take away from or to limit or restrict in any way the denominational rights of Roman Catholic Separate School Supporters and the Board under Section 93 of the Constitution Act, 1867.

15.12 - Certified Teachers

Where the *Education Act* permits employers to employ persons without teaching certificates to provide services previously required to be provided by certified teachers, the employer agrees to continue to employ certified teachers to provide such services, where the core duties of any position involve those duties normally performed by teachers. Notwithstanding the above, para-professionals may provide support services to teachers and students.

15.13 - Temporary Administrative Assignments

- (1) The Board may assign to a teacher the duties of an Administrator (Principal/Vice-Principal) for a temporary period of time not to exceed the balance of the current semester.
- (2) Any extension of an individual teacher's placement in such assignment may only be made with the approval of the Local Bargaining Unit.
- (3) No teacher shall be assigned the duties of an administrator without his/her consent.

- (4) Acceptance by the teacher of such temporary duties shall not interrupt the accumulation of seniority rights under the provisions of this collective agreement.
- (5) All other provisions of this collective agreement shall apply to the teacher during such period of temporary assignment, including the deduction and payment of OECTA dues.
- (6) Any teacher assigned such duties shall not participate in the evaluation of any member of this bargaining unit.
- (7) The Board agrees to replace a teacher who accepts a Temporary Administrative Assignment for a period of time as described in 15.14(1) and (2) with an occasional teacher.

15.14 - Replacing Teachers

- (1) Subject to Article 11:07, a recall teacher shall be hired to fill a vacancy arising due to the death of a teacher during the school year.
- (2) Where a teacher's absence is known to be for a full semester or school year, then a teacher shall be hired.

15.15 - Union Representatives

The board recognizes the appointment of one Association representative as Staff Representative at teach school and/or board site.

15.16 - Trial Administrative Assignment

- (1) Any teacher who chooses to apply for a vacant administrative position will have the right to return to a teaching position subject to Article XI, for a period not to exceed two (2) semesters. It is agreed that the teacher may not engage in any administrative duties that would involve evaluation of another teacher.
- (2) A teacher who so wishes may resign from the administrative position with two (2) weeks' notice.

15.17 - Access to Information - Association

(1) The Board shall attempt to provide by October 31 but no later than November 30

and by March 10 but no later than April 10 to the President of the Local Bargaining Unit, the qualifications, experience, annual salary and manner of calculation of the annual salary of each Teacher.

- (2) <u>a)</u> By September 30 but no later than October 31 of each year, the Board shall attempt to provide the President of the Local Bargaining Unit with an up-to-date directory containing the school location, home mailing address and home phone number for each member covered by this Collective Agreement. An attempt will be made to provide an update to this list by October 31 and March 30.
 - b) The Association will use the employees' personal information for bona fide purposes within its duty as the collective bargaining representative of the employees and it will act as the custodian of the employees' interests.

15.18 - Home Instruction

The Parties agree that job assignments for Home Instruction shall be allocated according to the Administrative Procedure & Guideline (APG) developed in consultation with OECTA Sudbury Secondary Local in 2008 and that any future revisions to this APG will be developed in consultation with the OECTA Sudbury Secondary Local.

Notwithstanding the above, the parties agree to the following:

- 1. Job assignments for Home Instruction for secondary school students shall first be offered to qualified members of the Secondary Local Bargaining Unit prior to qualified members of other OECTA bargaining units and to members of the Secondary Local Bargaining Unit who do not hold qualifications prior to any other unqualified candidates outside the Secondary Local Bargaining Unit.
- 2. The Home Instruction rate of pay for a member of the Secondary Local Bargaining Unit will be same as the Continuing Education Teacher rate, as defined in article 15.07 (3).

In witness whereof, the parties hereto have caused this Agreement to be signed in their respective name by their respective representatives thereunto duly authorized as of this 19th day of November, 2008.

Local	Bargaining Unit
Paula Peroni Chairperson	Dan Charbonneau OECTA Secondary
Catherine McCullough Director of Education	

SCHEDULE "A"

Effective September 1, 2008

Year	A1	A2	A3	A4
0	40274	41720	16096	49722
0	40374	41720	46086	48732
1	42763	44435	49184	52174
2	45151	47151	52280	55614
3	47539	49867	55376	59056
4	49928	52584	58472	62496
5	52316	55300	61568	65938
6	54704	58016	64665	69380
7	57093	60732	67762	72821
8	59481	63447	70858	76261
9	61870	66163	73954	79703
10	64259	68879	77051	83144
11	66646	71595	80147	86585

Effective September 1, 2009

Year	A1	A2	A3	A4
0	41585	42972	47469	50194
1	44046	45768	50660	53739
2	46506	48566	53848	57282
3	48965	51363	57037	60828
4	51426	54162	60226	64371
5	53885	56959	63415	67916
6	56345	59756	66605	71461
7	58806	62554	69795	75006
8	61265	65350	72984	78549
9	63726	68148	76173	82094
10	66187	70945	79363	85638
11	68645	73743	82551	89183

Effective September 1, 2010

Year	A1	A2	A3	A4
0	42833	44261	48893	51700
1	45367	47141	52180	55351
2	47901	50023	55463	59000
3	50434	52904	58748	62653
4	52969	55787	62033	66302
5	55502	58668	65317	69953
6	58035	61549	68603	73605
7	60570	64431	71889	77256
8	63103	67311	75174	80905
9	65638	70192	78458	84557
10	68173	73073	81744	88207
11	70704	75955	85028	91858

Effective September 1, 2011

Year	A1	A2	A3	A4
0	44118	45589	50360	53251
1	46728	48555	53745	57012
2	49338	51524	57127	60770
3	51947	54491	60510	64533
4	54558	57461	63894	68291
5	57167	60428	67277	72052
6	59776	63395	70661	75813
7	62387	66364	74046	79574
8	64996	69330	77429	83332
9	67607	72298	80812	87094
10	70218	75265	84196	90853
11	72825	78234	87579	94614

School Year

This will confirm the Board's agreement that the school year will not exceed 194 days for the duration of this Collective Agreement.

In consultation with the other School Boards with which the Board shares transportation services, every effort will be made to arrange for a turn around day at the beginning of the second semester.

Paula Peroni Chairperson Sudbury Catholic District School Board

Posting Process for New Schools

This will confirm the Board's agreement to advance the time of posting of vacant positions, once identified, to provide for a greater opportunity for teachers to apply for positions in new schools.

Paula Peroni Chairperson Sudbury Catholic District School Board

Teachers Assigned to Non-Credit Programs

Teachers assigned to non-credit programs shall retain their assignments, subject to the and Article XI (Staffing Process).	s at the date of this Collective Agreement e operation of Article IX (Staffing Formula)
Paula Peroni Chairperson Sudbury Catholic District School Board	Dan Charbonneau President O.E.C.T.A. Secondary Teachers' Local Bargaining Unit

Adult Continuing Education Program

The parties agree that procedures, guidelines and compensation shall be determined in consultation with OECTA Secondary with specific reference to the assignment of any marking of Continuing education Correspondence Credit Courses, including Independent Learning Courses to Adult Continuing Education Teachers

Paula Peroni Chairperson Sudbury Catholic District School Board

<u>It is agreed that access will be provided to re</u>	<u>elevant Board Administrative Procedures</u>
and Guidelines (APG).	
, , ,	
Doule Denoni	Don Charleson and
Paula Peroni	Dan Charbonneau
Chairperson	President
Sudbury Catholic District School Board	O.E.C.T.A. Secondary Teachers' Local
	Bargaining Unit

E-Learning Credit Courses

Prior to the implementation of e-learning there will be a committee formed with equal representation of both parties to discuss safe guards to the teachers' working conditions.

Paula Peroni Chairperson Sudbury Catholic District School Board

Best Practices

The Superintendent of Employee Relations will meet with all of the Secondary School Principals to explore best practices. Prior to the meeting, the Superintendent of Employee Relations will consult with the President of OECTA Secondary to set agenda items that will be addressed with the Principals. Agenda items will include but are not limited to:

- -Credit Rescue
- -Credit Recovery
- -Credit Integrity
- -E-learning -Program Leader consultation
- -Posting process

After the meeting with the principals has occurred, the Superintendent of Employee Relations will meet with the Unit President to discuss the best practices agreed upon.

Paula Peroni Chairperson Sudbury Catholic District School Board

Dan Charbonneau President O.E.C.T.A. Secondary Teachers' Local Bargaining Unit Local Bargaining Unit

LETTER OF UNDERSTANDING

No Hiring of Non Members

The Board will not hire any additional comembers of the Ontario English Catholic To	
Paula Peroni Chairperson Sudbury Catholic District School Board	Dan Charbonneau President O.E.C.T.A. Secondary Teachers' Local Bargaining Unit

Letter of Understanding

Staffing (PDT)

The parties agree that the Joint Board Staffing Committee's terms of reference shall include the following:

1. Review and share the allocation of staff hired by the Board for the expansion of secondary school programming in accordance with the new allocation in the Grants for Student Needs.

September 1, 2008: 0.19 teacher per 1,000 secondary pupils;
September 1, 2010: 0.70 teacher per 1,000 secondary pupils;
September 1, 2011: 1.02 teacher per 1,000 secondary pupils;
August 31, 2012: 1.35 teacher per 1,000 secondary pupils.

- a) The Board will hire the full complement of additional funded secondary teachers that result from this new allocation; for the 2008-2009 school year, the hiring of additional teachers shall occur at the second semester.
- b) The JBSC will be engaged in the development of a strategy to expand secondary school programming which may include increases in course offerings and strategic class size reductions, and in the allocation of this additional staffing.

Paula Peroni Chairperson Sudbury Catholic District School Board

Letter of Understanding

PDT Benefits and Enhancements

All Group Benefits shall remain status quo for 2008 through August 31, 2010. Conditional upon approval by the lieutenant Governor-in-Council, the parties will meet by January 15, 2010 to determine the allocation towards benefit coverage, improvements funded through the OECTA Secondary Local Bargaining Unit's share of the benefit enhancement funding in accordance with the Provincial Discussion Table Agreement to be implemented on September 1, 2010.

<u>Upon request, the Board shall provide the OECTA Secondary Local Bargaining Unit with the requested disclosure to inform decision making on this matter. The nature of the disclosure shall include, but not be limited to, any information provided by the Board in a public procurement process.</u>

The OECTA Sudbury Secondary Local Bargaining Unit is committed to educating its members on mutually agreed-to-cost-effective practices regarding the Benefit plans as outlined in Article VI in consultation with the Superintendent of Education & Employee Relations and the Manager of Human Resources.

Paula Peroni Chairperson Sudbury Catholic District School Board

Letter of Understanding

Member Education on Leaves

The OECTA Sudbury Secondary Local Bargaining Unit is committed to educating its members on teacher wellness and the appropriate use of sick leave under Article 6.06 and personal leave under Article 12.02 through general meetings and an OECTA meeting to be delivered by the President of the Bargaining Unit at each secondary school.

Paula Peroni Dan Charbonneau Chairperson President

Sudbury Catholic District School Board O.E.C.T.A. Secondary Teachers' Local

Bargaining Unit

Appendix I

(Compensation and Health Benefits Attachments) OECTA SECONDARY TEACHERS

Enclosed you will find a copy of the current collective agreement between the Board and the OECTA

Secondary Teachers' local bargaining unit. This document outlines compensation and health benefits as well as other conditions of your employment as a Secondary teacher with the Sudbury Catholic District School Board.

Please review the enclosed summary of the coverage provided by the various benefit plans, as well as Article VI - BENEFITS in the collective agreement.

If you wish additional information or clarification in the matter of benefits entitlement, please contact:

1. Cowan (re: group life, extended health and dental)
Telephone: 1-888-509-7797 Website Info: www.cowangroup.ca

2. Ceridian (Employee and Family Assistance Program)
Telephone: 1-877-207-8833 Website Info: www.lifeworks.com

3. OTIP (Long Term Disability) OECTA local unit contact, telephone: 521-0839

4. The Sudbury Catholic District School Board's benefit clerk, telephone: 673-5620

The complete information that you are required to provide is the following:

- 1. Life Insurance beneficiary designation (mandatory for all employees)
- 2. Health care benefits information (optional for part-time employees) ...all inclusive of: Extended Health, Drug plan, Vision and Semi-private
- 3. Dental benefits (optional for part-time employees)
- 4. Voluntary Accidental Death & Dismemberment Insurance (optional for all employees)
- 5. Long Term Disability Insurance (mandatory for all employees)
- 6. Optional Life Insurance
- 7. Positive Enrolment Form (Part 'A' and 'D' are mandatory; Parts 'B' and 'C' depend on the status of each individual)

Please ensure to contact the Benefits Clerk at 673-5620 if any of the following events occur during your employment:

- Marital status change
- Beneficiary change
- Coordination of benefits change
- Change in dependants (e.g. birth, death, over-age student)

You have access to pay information, benefit coverage, and dependant information on the Board's website, through the Intranet. Please review your information on a regular basis to ensure information is accurate and current.